Seton Hall University

eRepository @ Seton Hall

Seton Hall University Dissertations and Theses (ETDs)

Seton Hall University Dissertations and Theses

Spring 2-2-2023

Exploring Principals' Perceptions about Collective Efficacy in Urban K-8 New Jersey Schools Designated for the School **Improvement Process**

Crystal M. Joye joye.crystal@gmail.com

Follow this and additional works at: https://scholarship.shu.edu/dissertations



Part of the Educational Leadership Commons

Recommended Citation

Joye, Crystal M., "Exploring Principals' Perceptions about Collective Efficacy in Urban K-8 New Jersey Schools Designated for the School Improvement Process" (2023). Seton Hall University Dissertations and Theses (ETDs). 3065.

https://scholarship.shu.edu/dissertations/3065

| Exploring Principals' | Perceptions about Collective Efficacy in Urban K-8 New Jersey Schools |
|-----------------------|---|
| | Designated for the School Improvement Process |

Crystal M. Joye

Submitted in partial fulfillment of the requirements for
The Degree of Doctor of Education

Department of Education Leadership, Management & Policy

Seton Hall University

2023

© 2023 Crystal M. Joye

All Rights Reserved



COLLEGE OF EDUCATION & HUMAN SERVICES
DEPARTMENT OF EDUCATION LEADERSHIP MANAGEMENT & POLICY

APPROVAL FOR SUCCESSFUL DEFENSE

Crystal M. Joye has successfully defended and made the required modifications to the text of the doctoral dissertation for the **Ed.D.** during this **Spring** Semester.

DISSERTATION COMMITTEE

(please sign and date)

| Dr. David Reid | |
|------------------|------|
| Mentor | Date |
| Dr. Al Galloway | _ |
| Committee Member | Date |
| Dr. Roger Ball | |
| Committee Member | Date |

The mentor and any other committee members who wish to review revisions will sign and date this document only when revisions have been completed. Please return this form to the Office of Graduate Studies, where it will be placed in the candidate's file and submit a copy with your final dissertation to be bound as page number two.

Abstract

Exploring Principals' Perceptions about Collective Efficacy in Urban K-8 New Jersey Schools

Designated for the School Improvement Process

Crystal M. Joye

Leadership plays an important role in determining organizational culture and maintaining a commitment to the collective goal. Leaders not only play a critical role in developing and implementing policies to direct an organization toward a goal, but also serve as role models for those with whom they work. Research has shown that relationships between principals and teachers are critical to impact teaching practices, school climate, and students positively, yet these relationships can affect student learning and achievement.

Research has also shown that students' overall performance in school and any decrease in their overall achievement data are directly related to teacher self-efficacy (Hoy & Hoy, 2009). Prior to the designation of underperforming, a school's data indicated academic deficits for 3 years or more. The public identifier as targeted and/or comprehensive may have further implications of reduced teacher and principal efficacy.

The purpose of this qualitative narrative study was to explore principals' perceptions about collective efficacy in urban K-8 schools in New Jersey designated for the school improvement process. Principals' responses to the research question and sub-questions guided this study:

- 1. What are principals' perceptions of community collective efficacy as a means towards school improvement?
 - a. How do principals describe the value/possibility of community collective efficacy?

b. How do principals describe the challenges of community collective efficacy?

Collective efficacy in schools identified as in need of improvement is critical to enhancing student achievement. Significant findings revealed that collective efficacy contributes to improved student outcomes and can be fostered by effective leadership strategies. Teacher collective efficacy has also been shown to have a positive effect on the school environment for all parties involved in the educational process. Recommendations for districts included implementation of professional development, research-based and targeted interventions, and efficacious practices directly related to school improvement. Policy recommendations are relevant to teacher and leadership programs, including effectiveness and cultural programs.

An avenue of future research is the investigation of whether increasing the DEI initiatives of K-12 schools leads to differences in student self-efficacy and teacher self-efficacy, which would, in turn, impact collective efficacy. An interesting concept is the idea that collective efficacy promotes inclusion. Recommendations for practice include an early warning indicator evaluative structure to support principals and schools at risk of being identified as in need of improvement.

Keywords: efficacy, self-efficacy, comprehensive school, failing schools, principal perception, data-driven, school improvement, stigmatized, underperforming, school culture

Acknowledgments

I would like to thank my committee members, Dr. David Reid, Dr. Roger Ball, and Dr. Albert Galloway, for agreeing to part of my journey. Your thoughtful feedback pushed me to think critically and reflect. To my advisor, Dr. David Reid, thank you for helping me stay focused on the finish line. I was able to distill and discern well-meaning advice and navigate my path. I appreciate your guidance, patience, and positive words.

Many thanks to my colleagues and friends-turned-family from Cohort 23. We made it through a pandemic and accomplished the unthinkable as we learned to navigate virtual learning for the first time. To Tenelle and Steven, "the SHU family," thank you for your friendship, honesty, ability to be straightforward, and unapologetic nature when it came to the work. Our daily accountability calls, Zoom study sessions, and meet-ups kept me accountable and focused on the goal. Seton Hall truly brought us together and we are bonded for life!

To the Off School Grounds Family (OSG), Dr. Dennis McKesey, CEO, my sister and long-time friend April Leong, and the ladies of OSG, you are an amazing group of educators who poured into me during a time of need. I am in a better place, both personally and professionally, because of your words of encouragement, morning messages, the Thursday meetup sessions, and your continued support. Thank you to everyone who gave their time to participate in the research portion of my study.

A special thank you to the ladies of Zeta Phi Beta Sorority, Incorporated, The Groundbreaking Gamma Alpha Pi Zeta Chapter. I am also incredibly grateful to my sisters, my Sorors, for cheering me on every step of the way. Thank you for giving me grace as I juggled multiple tasks and responsibilities. To Dr. Courtney Johnson and Antinea Middleton, thank you

for reminding me to maintain a balance in life. Your unwavering support is appreciated. #Iron sharpens Iron.

Thank you to my mother, Lynett, and my grandmother, Sylvia, for always encouraging me to follow my dreams. Mom, you inspire me to be the best at whatever God puts before me. Thank you for being my person, my rock. To my baby brother, Jeffery, I hope to continue being a role model and inspiration to you.

To my husband, William, and my girls, Deneja, Teja, and Laney, thank you for sharing this experience with me from start to finish. Each of you fully understood my why and the importance of this journey. I want you to know that without you, I would not have made it through the late nights on my laptop, the summer sessions at SHU, Statistics class, and the many sacrifices each of you made to support me.

In closing, I want to thank my extended family: my mom-in-love Yvonne, sisters Kathryn, and Tara, for reminding me of my worth. I knew that whatever my dissertation was, it would be full of joy(e). I also knew that it needed to highlight the change agents that we are, our purpose in education, and our service to others as leaders. Thank you for always rooting for me.

This huge accomplishment is also dedicated to my late father Jeffery, and grandparents, Joseph and Mary Lee Simmons. Thank you for watching over me. I know you are all very proud.

C. M. J.

Table of Contents

| Abstract | iv |
|--|------|
| Acknowledgments | vi |
| List of Tables | xiii |
| CHAPTER 1: INTRODUCTION | 1 |
| Background of the Problem | 1 |
| Statement of Problem | 5 |
| Research Questions | 6 |
| Statement of Purpose | 7 |
| Significance of the Study | 8 |
| Theoretical Framework | 9 |
| Study Design and Methodology | 10 |
| Limitations and Delimitation | 12 |
| Definitions of Terms | 13 |
| Summary | 15 |
| CHAPTER 2: LITERATURE REVIEW | 17 |
| Introduction | 17 |
| Literature Review | 17 |
| National Student Achievement Standards | 17 |
| Targeting Underperforming Schools | 19 |
| School Improvement Process | 20 |
| Difficulties Faced by Urban Schools | 22 |
| Student Self-Efficacy | 24 |

| Teacher Self-Efficacy | 25 |
|--|----|
| Professional Development | 27 |
| School Collective Efficacy | 27 |
| Principal and School Collective Efficacy | 28 |
| Theoretical Framework | 30 |
| Social Cognitive Theory | 30 |
| Mindset Theory | 31 |
| Gaps in the Literature | 32 |
| Summary | 33 |
| CHAPTER 3: METHODOLOGY | 35 |
| Introduction | 35 |
| Purpose Statement and Research Questions | 35 |
| Research Design | 36 |
| Data Collection | 37 |
| Research Protocol | 38 |
| Participants | 40 |
| Semi-Structured Surveys | 41 |
| Saturation | 43 |
| Reliability and Validity | 43 |
| Limitations | 44 |
| Summary | 45 |
| CHAPTER 4: RESULTS | 46 |
| Introduction | 46 |

| Data Analysis and Results | 47 |
|---|----|
| Subsection One: Principals' Perceptions of Community Collective | |
| Efficacy as a Means Towards School Improvement | 48 |
| Principals' Definitions of Collective Efficacy | 48 |
| Collective Efficacy as an Ability | 48 |
| Collective Efficacy as Collaboration and Teamwork | 49 |
| Collective Efficacy as Work Towards a Positive Goal | 50 |
| Current State of School Collective Efficacy | 51 |
| High Collective Efficacy | 51 |
| High Collective Efficacy, Then Low Collective Efficacy | 53 |
| Low Collective Efficacy | 53 |
| Utilizing Collective Efficacy for Identifying Interventions | 54 |
| Data-Driven Process | 54 |
| Response-to-Intervention (RTI) | 55 |
| Team-Driven Interventions | 56 |
| Role of Collective Efficacy in Strategic Management | 56 |
| Data Analysis and SMART Goals | 57 |
| Expanding Planning Network | 58 |
| Subsection Two: Perceived Benefits of Collective Efficacy | 60 |
| Role of Collective Efficacy in Decision-Making | 60 |
| Principals Make Goals with Efficacy in Mind | 61 |
| Team Orientation | 62 |
| Truth and Transparency | 64 |

| Current State of School Collective Efficacy: Perceived Benefits | 64 |
|---|----|
| Collective Efficacy Results in Positive Effects | 65 |
| Principals Generally Make Purposeful Decisions Based on | |
| Perceived Collective Efficacy | 66 |
| Yes, Purposeful Decisions with Collective Efficacy Enhanc6 | |
| Team Orientation | 66 |
| Yes, Purposeful Decisions with Collective Efficacy Maintain | |
| Morale | 68 |
| Subsection Three: Challenges Associated with Collective Efficacy as a | |
| Means Towards School Improvement Efforts | 68 |
| Current State of School Collective Efficacy: Perceived Challenges | 68 |
| Mindset Shift | 69 |
| Work in Progress | 69 |
| Impact of Needs Improvement Rating in Collective Efficacy | 70 |
| Needs Improvement Rating: Negative Effects in Schools | 72 |
| Assumption That Parents and School Staff Do Not Care | 72 |
| Lack of Enrollment and Lack of Retention | 72 |
| Lack of Trust in School Administration | 73 |
| No, Collective Efficacy Hinders Decision Making | 74 |
| Collective Efficacy Does Not Drive Decision-Making | 74 |
| Negative Perceptions | 75 |
| Impact of Needs Improvement Rating | 77 |
| Summary | 78 |

| CHAPTER 5: RESULTS AND DISCUSSION | 79 |
|--|-----|
| Summary | 79 |
| Contextual Analysis of Themes | 79 |
| Research Question 1 | 80 |
| Influence of Collective Efficacy on Decision-Making, Strategic | |
| Planning, and Identifying Interventions | 80 |
| Sub-question 1a | 82 |
| Mindset for Change, or Growth Mindset | 82 |
| Increased Collective Efficacy, Increased Inclusion | 84 |
| Sub-question 1b | 85 |
| The Needs Improvement School Rating | 85 |
| Collective Efficacy Can Hinder Effective Decision-Making | 86 |
| Limitations | 87 |
| Implications for Future Practice | 88 |
| Suggestions for Future Research | 90 |
| Summary | 90 |
| References | 92 |
| Appendix A: Codebook | 112 |
| Appendix B: IRB Approval | 125 |
| Appendix C: Informed Consent Form | 126 |

List of Tables

| Table 1 | Demographic Characteristics of the Participants | 41 |
|---------|---|----|
| Table 2 | Principals' Definitions of Collective Efficacy | 48 |
| Table 3 | Principals' Perceptions of the State of School Collective Efficacy | 52 |
| Table 4 | Principals' Perceptions of Process for Identifying Interventions | 54 |
| Table 5 | Principals' Perceptions of Role of Collective Efficacy in Strategy Planning | 57 |
| Table 6 | Principals' Perceptions of How Collective Efficacy Affects Their Decision-Making. | 61 |
| Table 7 | Principals' Decision-Making Is Generally Based on Collective Efficacy | 67 |
| Table 8 | Principals' Perception of the Schools' Needs Improvement Rating | 71 |

CHAPTER 1: INTRODUCTION

Background of the Problem

One of the most enduring challenges facing schools and those who study them regards how to improve learning while also closing persistent achievement gaps. When teams have a shared belief in collaborative efforts, researchers have found these challenges can be overcome, at least in some respects (Wilcox & Lawson, 2018). When teachers at school believe in collective ability to influence student performance, they become more creative and productive. Bandura found this interesting pattern of human behavior as "self-efficacy" (Voelkel & Chrispeels, 2017). It is defined as a group's collective belief in the group's collective ability to organize and implement the practices necessary to achieve a certain level of success (Wilcox & Lawson, 2018). Self-efficacy is an individual's self-awareness of the ability to learn, perform, or receive a particular task.

Donohoo and Katz (2017) found a strong positive correlation between student efficacy and achievement. Several studies have confirmed the relationship between positive self-efficacy and student performance. The Bandura Framework for Social Cognitive Theory states that teachers are role models for student learning. Learning is the process of repeating what the student sees in the (teacher's) model and is enhanced by an interactive experience. Previous research has emphasized a consistent relationship between performance and achievement learning by all ages, genders, races, learning environments, and socioeconomic statuses. Studies have shown that there is a positive correlation between teacher and student self-efficacy and the level of commitment to achieve the goals and objectives involved in the school improvement process (Cansoy & Parlar, 2018; Prenger & Schildkamp, 2018). Teachers have also shown that they are obsessed with overcoming obstacles and challenges faced during the high-level school

improvement process. It has produced a higher level of overall effectiveness at school that improves academic performance throughout the school community.

Labeling poorly performing people alters educational and learning behaviors and leads to negative implications associated with interpreting scores that indicated poor performance. Of further concern is the potential misconception among the school community that poor grade labels are a direct measure of student performance as evaluated by schools. Other indicators that viewers do not necessarily associate with poor score scales may include situations such as (a) schools that rate less than 95% of students enrolled at the beginning of the semester; (b) lack of annual progress in smaller subgroups of students; and (c) number of English Language Learners (Wilcox & Lawson, 2018).

Other aspects that prevent attendance at school, especially in middle and high school classrooms, include displacement, homelessness, parenting children, guardianship of younger siblings, and family members (Ni et al., 2018). All factors negatively affect attendance and encourage disciplinary action that promotes poor academic performance. Problems that can adversely affect school scores occur at a disproportionately high level in schools in urban and geographic areas affected by risk factors. These problems include rising poverty rates, homelessness, crime, abandonment of education, teen pregnancy, and parental involvement to drop out of school. Schools serving disadvantaged students are often forced to improve on factors unrelated to school-based education. Lower asset values within a school's proximity can prolong the cycle of poor grades due to the public's lack of awareness of the quality of school education (Ninković & Knežević, 2018). Depleted property values due to low-performance scores can facilitate student enrollment that is disproportionately biased because of its low socioeconomic background.

The surrounding poor school community needs to interpret and navigate the negative implications associated with poor school performance labeling. Widespread opportunities for school selection and negative perceptions within the surrounding community can affect the level of neighborhood support that poorly performing schools receive when initiating and mediating the school improvement process (Zheng et al., 2019). Schools with already low levels of self-efficacy can be even less efficient. Such levels hinder the successful implementation of school improvement and strengthen the negative perceptions of the public.

Teacher performance is closely linked to the perception of overcoming the everyday challenges and pressures faced when performing daily tasks in the classroom. Teacher responsibilities are more diverse because they facilitate each student in succeeding within the expectations of a series of learning lessons. Bandura's Social Cognitive Theory identified experience as a factor and promoted efficiency (Zheng et al., 2019). Effective teachers recreate success from previous successful classroom experiences. Strengthening teachers helps them overcome challenges. Schools who have low self-efficacy teachers face and should address the additional needs and stresses that teachers have when working in such a school environment. However, this may not always be possible; as a result, the success of school improvement efforts becomes even more difficult.

Schools with traditionally poor grades have higher employee turnover rates than other schools. Schools that are considered poor performers in the community face challenges in attracting and retaining working teachers (Voelkel & Chrispeels, 2017). While developing the specialized expertise needed to support student performance, union agreements and school board policies for low-level employees are a fruitful new school appointment intended for teachers.

Lack of contact with the school community impedes the development of self-efficacy that teachers need to become educational leaders in their classrooms.

Studies have shown a positive relationship between students with a high level of self-efficacy and students with a high level of academic performance (Donohoo & Katz, 2017; Prenger & Schildkamp, 2018). They have also shown a positive relationship between teachers with a high level of self-efficacy, student success, and teachers with high student achievement. Bandura confirmed this relationship when he identified a positive relationship between teacher self-efficacy and student performance (Voelkel & Chrispeels, 2017). Other studies have identified a triangular effect between levels of self-efficacy levels of teachers, high levels of student self-efficacy, and higher-level achievement.

Evidence of the potential impact of a negative relationship between self-efficacy and student performance during the school improvement process is not widespread. Since the founding of No Child Left Behind (NCLB), schools have been labeled as having poor grades and designated as disadvantageous by the public (Reaves & Cozzens, 2018). Government agencies are free to design state labeling systems using terms that imply poor performance, such as *failure* or the actual term *poor performance* itself. Use of personal data sources leads to the expansion of policy outcomes from both individual and collective policymakers and may affect the self-efficacy of affected members. School teachers who are affected by the school community must work within boundaries and perceptions with unacceptably low grade scores. The challenges faced by school leaders during the school improvement process are heightened by the impact of low academic performance scores on climate and culture, and the overall effectiveness of the school community. Additional challenges are expected as similar processes continue within the framework for assessing social and environmental systems.

Much research exists on self-efficacy and its influence on these challenges. However, less research has used collective efficacy as a frame with which to view how individuals who work in schools move to overcome challenges. Collective efficacy is defined as a group's shared belief in its conjoint capability to organize and execute the courses of action (Bandura & Walters, 1977, p. 477). Like self-efficacy, it is shaped by four major sources: mastery experience, vicarious experience, verbal persuasion, and affective state (Bandura & Walters, 1977; Goddard et al., 2004); these are detailed further in Chapter 2.

Statement of Problem

Through my research and analysis, I sought to explore collective efficacy in urban K-8 schools in the state of New Jersey. I designed this qualitative study to understand the principal's school-wide use or failure to use collective efficacy strategies during the school improvement process.

The research problem I sought to understand was the use of collective efficacy and its impact on urban schools under review in the state of New Jersey. The role of collective efficacy was explored for whether principals' perceptions and the use of collective efficacy within their school communities have an influence on the decisions they make—specifically, the implementation of academic intervention strategies to improve practice and school-wide professional development plans during the school improvement cycle process.

Thus, I investigated if principals in urban K-8 schools designated as in need of improvement commonly face such issues as lack of strategies to improve instructional outcomes, selection of appropriate interventions, and barriers in the classrooms. According to Elmore (2008) and Mulford (2003), the prerequisite skills for functional school leadership require the

principal to engage in organizational learning and leadership to build capacity among all constituents in the school community through collective learning.

I also wished to explore further if the perception of principals differed among faculty members in urban K-8 New Jersey schools. Perceptions of principals varied among faculty members, contingent on such factors as faculty tenure status, years of experience in education, or time spent at a dedicated site. According to Bandura (1993), previous research has demonstrated that collective efficacy positively predicts students' academic achievement (also see Goddard et al., 2004). Current research has not addressed if collective efficacy is a factor in the reduction of inequities to close the achievement gaps in urban New Jersey schools.

This study will add to the current body of literature on the topic of collective efficacy.

The framework used for this research study; Bandura's Social Cognitive Theory (SCT), is a conceptual framework that underlies qualitative research (Reaves & Cozzens, 2018). The secondary conceptual framework for this study was Carol Dweck's theory of mindset (Linares, 2019). For this research, I considered the role of collective efficacy during the school improvement process and the principals' implementation of strategies and best practices that support school improvement. Thus, I sought to identify considerations for collective efficacy and how principals' thoughtful choices and implementation of school-wide structures can align with the school improvement process.

Research Questions

The following research questions and sub-questions were used to guide this study:

RQ1: What are principals' perceptions of community collective efficacy as a means towards school improvement?

SQ1a: How do principals describe the value/possibility of community collective efficacy?

SQ1b: How do principals describe the challenges of community collective efficacy?

Statement of Purpose

The purpose of this qualitative narrative study was to explore urban principals' perceptions of collective efficacy and the role it has played in their journey to implement strategies and structures during the school improvement process in New Jersey. This study investigated aspects of professional development and embedded intervention strategies either enacted or not enacted during the school improvement process. The study further determined if principals' perceptions of collective efficacy influenced them in their overall decision-making. Lastly, this study explored if collective efficacy and school-wide efficacious perceptions of leaders and teachers varied among staff members, contingent on years of service in education, tenure status, and the number of years employed at an urban school location in New Jersey.

Principals serve as the driving force in the school improvement process. This study can add to the existing body of literature on collective efficacy as it relates to the school improvement process in urban New Jersey schools.

More specifically, this study examined principals' perceptions of collective efficacy and how these perceptions influenced their work of school improvement. The process of classifying the school as a poor-grade school is, therefore, the goal of the school improvement process. However, the overall assignment as a school with low-grade scores in the school improvement process can undermine the self-efficacy of the school community (Ni et al., 2018). Self-awareness within the school community of poor-performing challenges can have a negative

impact that can impede behavior and sustainable progress in improving the school. Fluctuations in the level of teacher self-efficacy can change over time during the school's improvement process (Boyce & Bowers, 2018). To complete the school's improvement process, the school must demonstrate its ability to use indicators to measure academic success. The scale contains information about student grades, as measured using metrics set by relevant policymakers.

Studies have shown that increased student grade data should be associated with increased levels of teacher self-efficacy. Principals who lead the school improvement process need to understand the effectiveness and impact the process has on academic performance.

Significance of the Study

Each state has the option of including additional school quality indicators under the Elementary and Secondary Education Act (ESSA, Every Student Succeeds Act) to assess school performance and responsibilities (Wilcox & Lawson, 2018). This applies to all students and subgroups, including post-higher education preparations. Student participation or school atmosphere grades may be constructed as indicators such as school climate, but ESSA does not address this directly.

Interventions in schools that are considered poor include increased opportunities for professional development at the school level. Such professional development is based on each teacher's minimum ability level. Studies have shown that teacher performance influences attitudes and willingness to adhere to new educational practices (Donohoo & Katz, 2017). The present study provided a deeper understanding of resource allocation and the setting of standards and norms that guide school leadership. This understanding includes support systems and resources to help overcome undiscovered obstacles and unintended challenges facilitating a more efficient and manageable process for improving schools. The continued education and training of

principals help school and district leaders lead the school improvement process in understanding the importance and role of collective efficacy in achieving success. Policymakers and decision makers can gain a deeper understanding when allocating resources and setting standards for guiding school leadership. In turn, strategic preventive plans to educate the public can reduce the potential for negative perceptions of school abilities.

Theoretical Framework

Bandura's Social Cognitive Theory (SCT) is a conceptual framework that underlies qualitative research (Reaves & Cozzens, 2018). Bandura (1977) defined self-efficacy as people's belief in their ability to succeed. According to Bandura, students who have a higher level of selfefficacy are more successful than students who do not; the latter group was comparable and considered low self-efficacy. Students make continuous cognitive assessments of their learning experience, including observations, opinions, and personal beliefs to determine if their performance is effective within their abilities and/or controls. Bandura's research focused on the triangular interactions of personal, behavioral, and environmental factors that affect learning outcomes and performance. Students develop more personal and effective behavior if they succeed in aggressive strengthening to overcome challenges and pitfalls (Ninković & Knežević, 2018). Heuristic feedback that guides students' learning by clearly distinguishing between correct and incorrect learning behavior leads to effective results (Sun & Xia, 2018). Many opportunities in a learning environment to observe peers who are considered equal when striving to overcome similar challenges and succeed will enhance students' self-efficacy. This triangle of personal, behavioral, and environmental factors has the greatest impact on self-efficacy.

The second conceptual framework for this study was Carol Dweck's theory of mindset (Linares, 2019). Dweck's research showed that the ability of individuals to achieve results could

influence their perception of their ability to succeed (Maier et al., 2017). Fixed thinking imposes limits on a person's ability to achieve beyond the perceived limits of one's ability. The idea of growth pushes the boundaries of success through academic flexibility (Ford et al., 2019). The difference in learning attitudes between fixed and growing thinking complements students' willingness to tackle academic challenges. Predefined limits on fixed thinking lead to some skill acquisition. The idea of growth recognizes limits and enduring difficult situations with the determination to overcome adversity (Smith et al., 2018a). This present study determined whether school staff can determine the adequacy between grade data and teachers' effective attitudes during the school improvement process.

Study Design and Methodology

Implementing a school improvement process requires the ability of school administrators to assess or coordinate all aspects of the school community effectively. Understanding the self-efficacy of teachers, students, and parents facilitates a centralized decision-making process that leads to change in organizational management decisions, professional development, and resource allocation; building capacity for sustainable school improvement depends on maximizing professional growth and student performance (Lee & Louis, 2019). A low level of self-efficacy can undermine or disrupt administrative efforts to initiate timely changes that lead to the sustainable improvement of the school. School leaders continually look for the right resources as well as an exemplary professional growth model educational skills training that supports the diverse needs of teachers throughout the school improvement process. A support system that addresses the flaws identified in the school community is important for resolving issues that lead to poor academic performance (Donohoo, 2018). Low self-efficacy and group effectiveness can hinder the successful implementation of professional practices and actions that improve overall

performance, rely on an effective school community to improve consistent academic performance, and create and reinforce the willingness to design positive paths to improvement.

Long-term continuous improvement requires a cohesive and effective school community.

Information about school effectiveness is not a direct measure of the public school system. Limited information about the school atmosphere and culture contained in a school quality survey varies from school to school (Donohoo, 2018). It depends on the system to which one condition survey is assigned from region to region. As a result, for the present study, the qualitative approach was considered the most appropriate approach for conducting such research. The interviews were conducted by using a semi-structured process to gather the interviewees' perceptions through free-form, discussion-led questions. Voelkel and Chrispeels (2017) suggested that interview questions can be divided into six distinct categories according to the guidelines, given that the interviewer's intended experience and behavior can provide information about his or her specific experience and behavior. This qualitative study explored how principals' awareness of effectiveness played a role in the decision-making process during the school improvement process in New Jersey schools. This led to a great understanding of the effects of the principals' perceptions of effective change while leading the school improvement process.

Designating the overall school quality labeling process and the school improvement process influences research methods. Principals run schools that are officially considered inappropriate for inefficient classification and school improvement labels. Thus, the researcher of this study used a semi-structured interview approach to mitigate the principals' concerns and encourage participation through more engaging discussions about the school. The semi-structured interview method allowed the principals to reflect on their ideals and professional

experience in an unlimited manner as they work to improve the school. The researcher thus collected and analyzed qualitative data that informed how the school improvement decision-making process was associated with effective perceptions by New Jersey school municipal school principals. The technique of intentional sampling was used to investigate a particular group of principals who promoted the development of a school. Interviews were conducted specifically with selected principals who lead or have previously led part of the modern school improvement process and how their effective beliefs and perceptions influenced their leadership and decision-making process. The researcher interviewed 10 principals from urban K-8 elementary schools in New Jersey who responded to the school improvement process progress questionnaire.

Trilateral analyses of the encrypted field recorded professional development content and the difference in explanations of effectiveness between strategic intervention teachers. The coded data revealed subjects and patterns that matched the questions that led to the research framework. Analysis of the data was completed to determine if awareness of the principals' effectiveness played a role in the decision-making process during the school improvement process in urban New Jersey schools.

Limitations and Delimitations

Limitations may include research bias as the researcher is an educator and former principal of an urban school currently working in New York. As a Director of Student Support Services in a turnaround school, the researcher was fully aware of the possibility of researcher bias, given the topic of the study. Given the researcher's experience and understanding of the school improvement process through the lens of an educator, it was necessary to remain true to the IRB process and interview structure to safeguard the integrity of the study. This researcher

collected qualitative data from urban K-8 schools designated for the school improvement process in the state of New Jersey.

The principals of the selected schools were either currently engaged or had prior experience with leading a school identified as needing improvement. This study was designed to document the perceptions of 10 principals across urban schools in New Jersey. However, one cannot assume that the interpretation of the findings can apply to school districts outside those selected for this research study. Additionally, the findings were not related to grade ranges or school levels other than the K through 8 population.

The data collected from this research served as a 'snapshot' of time, reflecting the 10 principals' open and honest perceptions of the collective effectiveness of the schools designated for the improvement process. Professional experience, educational personality traits, and the school improvement process itself constituted the boundaries of the school administrators' perceptions. The researcher also brought personal perceptions of and experiences with the research topic being investigated. The researcher was also aware of the limited benefits of using semi-structured interviews with the principals, which did not allow for the researcher/interviewer to build rapport.

Definitions of Terms

This list provides the definitions of terms used in this study and related to concepts used in education.

Underperforming or Comprehensive schools - Schools performing at or below the fifth percentile of all Title 1 schools (Lee & Louis, 2019).

ESEA - Elementary and Secondary Education Act, formerly known as the No Child Left Behind Act (NCLB), was completed when the President Barack Obama signed the Every Student

Succeeds Act (ESSA) into law on December 10, 2015. Its main purpose is as a civil rights law to address the need for equal opportunities in education for all students, including students with disabilities (Lee & Louis, 2019).

ESSA Accountability - The accountability indicators in the ESSA State Plan measure the progress of schools as they prepare students for a successful life after high school. Stakeholders, including New Jersey parents and educators statewide, provided input into New Jersey's ESSA State Plan (Voelkel & Chrispeels 2017).

School Culture - School culture encompasses all the attitudes, expected behaviors, and values that impact how the school operates (Goddard et al., 2017).

NJSLA - The New Jersey Student Learning Assessment (NJSLA) is a statewide test administered in New Jersey to students in elementary, middle, and high school. Subjects tested include English Language Arts (ELA), Mathematics, and Science. Grades 3-8 sit for ELA and Mathematics tests, while Grades 5, 8, and 11 sit for the Science test. The aim of this statewide test is to measure the progress of students in satisfying the level of set academic standards (Donohoo, 2018).

Title 1 Schools - Schools that receive supplemental funding to educate low-income students are identified as having a significant percentage of students who are eligible for free or reduced-price lunch (Linares, 2019).

Collective Efficacy - The effectiveness of the community (school) in overcoming obstacles to success in a particular task (Donohoo, 2018).

Culture - Integrated programming that separates the characteristics of one group from another (Boyce & Bowers, 2018).

Efficacy - Ability to achieve a specific goal or skill.

School Administration - All assigned school administrators and administrators.

Self-Efficacy - The beliefs that people have about their ability to acquire or achieve skills.

Senior Faculty Members - Faculty with the longest tenure or teachers appointed to a single school.

Sense of Efficacy - Overall efficiency or self-efficacy.

Social Cognitive Theory (SCT) – Bandura's theoretical framework for learning through an integrated observational context of social factors (Maier et al., 2017).

Summary

An effective metacognitive process may not be particularly necessary for the school's improvement process. However, specific guidelines for improvement can guarantee simple and timely success results. Studies have shown that there is a positive correlation between teacher self-efficacy and student self-efficacy. The high academic performance of students is the result of positive and effective relationships. Studies have also shown a positive correlation between overall school performance and student performance. School principals need to lead their schools' improvement process and support the development of effective behaviors that keep students performing well. The overall evaluation of the school and its placement in the school improvement process can affect the level of effectiveness of the entire school community. A deeper understanding of the unintended consequences of the labeling process leads to more informative practices regarding the framework for publicly sharing findings. School naming and existing policies that inadvertently lead to negative public perceptions challenge school principals to drive their schools' improvement process. Understanding the full impact of policy

on the school improvement process gives school management a broader perspective on how to guide a successful journey.

This research study is divided into five chapters to provide the reader with the context of collective efficacy in urban New Jersey schools designated for the school improvement process. This Introduction chapter consisted of the foundational elements of the study. The background, purpose, significance, and theoretical framework as it relates to collective efficacy were outlined. Chapter 2 provides a literature review inclusive of the definition of collective efficacy, the theory behind collective efficacy, an overview of the collective efficacy framework, research on collective efficacy in urban schools, and the identification of gaps in the literature. Chapter 3 provides an overview of the research design used to capture 10 principals' perspectives of collective efficacy in underperforming urban schools. The study's methodology, design, selection of principal participants, and data collection and analysis are discussed. Chapter 4 presents the findings and results of this investigation. It also analyzes and interprets the data through descriptive findings tied to the research questions. Finally, Chapter 5 provides a summary of the research, identified limitations, connections of the findings to previous research, and suggested implications for future research and practice.

CHAPTER 2: LITERATURE REVIEW

Introduction

As described in Chapter 1, the purpose of this qualitative narrative study was to explore principals' perceptions of collective efficacy in urban K-8 schools designated for the school improvement process in the state of New Jersey. Chapter 2 now reviews key literature on this topic and explores performance effects that may be relevant to the school improvement process. It also reviews the study's theoretical framework, self-efficacy, collective teacher efficacy, collective school efficacy, and principal and school collective efficacy. Finally, this chapter provides a review of the literature on performance issues that can undermine or disrupt the school's improvement process.

The main theoretical framework that underpinned this research was Bandura's Social Cognitive Theory (SCT). This chapter outlines SCT and discusses its impact on educational and learning processes. This chapter also provides a brief survey of ideas to describe the potential impact of previous constraints on the ability to achieve. A successful school improvement process requires a school community that is confident in its ability to achieve results and overcome challenges (Goddard et al., 2017). The use of overt high-performance behaviors helps schools achieve academic performance. The SCT explains the impact of social behavior and interaction on academic performance. Academic performance contributes to an undefined range of awareness of the school community's ability to succeed.

Literature Review

National Student Achievement Standards

The No Child Left Behind (NCLB) Act was signed into law in 2001 by President George W. Bush in response to waning national academic competitiveness and widening achievement

gaps. The legislation served primarily as a means of holding schools responsible for student achievement. The intent was to bring more attention and assistance to underachieving students; failure to comply with the new requirements put schools at risk of losing federal Title 1 funding (NCLB, 2001). Among the requirements of the NCLB Act was to ensure proficiency of all students, as determined in part by the results of standardized testing. The legislation was intended to reduce the achievement gap by incentivizing schools to put more resources towards students in special education, English Language Learners (ELLs), and children belonging to other disadvantaged groups (NCLB, 2001). However, NCLB has been widely criticized for its overemphasis on testing and excessive regulations micromanaging the educational process, as well as for being costly and ineffective (Butzin, 2007; Guilfoyle, 2006; Simpson et al., 2004). In fact, in 2004, the Virginia legislature passed a resolution calling NCLB "the most sweeping intrusions into state and local control of education in the history of the United States" (Simpson et al., 2004, p. 68). The dependence on testing as the primary metric of school success and coupling of funding to this outcome placed an additional burden on teachers, leading to educational practices that neglect some of the important but intangible aspects of learning in favor of material on which students will be tested (Guilfoyle, 2006).

The NCLB Act was eventually replaced by the 2015 Every Student Succeeds Act (ESSA), which represents a significant advancement from NCLB, but nonetheless retains some of the drawbacks of its predecessor (Mathis & Trujillo, 2016). Under ESSA, schools are still identified as needing improvement largely via the results of standardized testing, but additional parameters measuring student achievement may be included in this metric, as determined by individual states (Mathis & Trujillo, 2016). This offered states more flexibility in determining how to evaluate schools, and specifically allowed for more comprehensive methods than those

used under NCLB, which were limited to standardized tests in reading/language arts and mathematics skills. Because of NCLB's narrow focus on these subjects, many schools eliminated supporting curriculum in favor of emphasizing those areas on which they would be tested, leading to diminished availability of art, music, and physical education and decreased emphasis on subjects such as science and history (Darling-Hammond et al., 2016). However, while ESSA opened up the possibility of states incorporating additional metrics into their evaluation of school, it did not require them to do so. Even today, there is insufficient consideration of other measurements of school success beyond quantitative academic performance, which represents a critical gap in the ability to evaluate schools and student success appropriately. Significantly, the student achievement gap has persisted under ESSA, with students of low socioeconomic status or from racial minorities demonstrating lower average test scores and grade point averages, as compared to wealthy and/or White students (Hung et al., 2020). ESSA still requires states to identify the "lowest-performing" 5% of schools; these districts, and any that fail to graduate more than one third of high school students, are required to complete a comprehensive support and improvement plan, wherein they describe their plan to address the identified inadequacies (Darling-Hammond et al., 2016).

Targeting Underperforming Schools

Among the major criticisms of NCLB and ESSA is the identification of schools as failing to meet certain standards of achievement. Under NCLB, there was considerable pressure for schools to meet certain testing standards at the risk of losing federal Title 1 funding. While these standards were largely determined at a federal level under NCLB, ESSA has shifted some of the responsibility for this to individual states (Hodge & Welch, 2016). While this has offered states more flexibility to address the unique needs of their student populations, it has also resulted in

several states lowering their proficiency standards from those set federally under NCLB. Thus, rather than actually reducing achievement gaps, these new standards often result in schools being able to claim proficiency, despite students remaining underserved. The National Education Policy Center further recognized that much of the performance gap can be attributed to income inequality, and schools tasked with educating children from disadvantaged demographics should receive increased funding to better serve this population rather than be punished for failing to meet a universal standard that may be unrealistic, given their resources (Mathis & Trujillo, 2016). Unfortunately, ESSA remains underfunded; while considered a step in the right direction, it is unlikely to address the underlying causes of school inequities adequately (Mathis & Trujillo, 2016).

School Improvement Process

After a school is designated as having failed to meet academic standards, the school is faced with the challenge of making improvements, with the expectation of meeting state-determined criteria within a period of no more than 4 years. Under ESSA, the "lowest-performing" 5% of schools are required to complete a comprehensive support and improvement plan. This not only requires schools to develop a strategy to improve student outcomes but also requires that this strategy be informed by the state's accountability system, identify specific resource inequities, and include evidence-based interventions (Darling-Hammond et al., 2016). These requirements have been criticized as being ambiguous, which may pose a difficulty for schools to address the requirements of ESSA appropriately in their efforts to improve (Herman et al., 2017). Some researchers have argued for the need for increased oversight by state and local education agencies with regard to implementation of school improvement plans (Dunn & Ambroso, 2019).

ESSA places a substantial burden for the school improvement process on school leadership, acknowledging principals as playing a key role in driving educational outcomes (Haller et al., 2016; Herman et al., 2017). Many of the evidence-based practices supported under ESSA relate to improving leadership strategies, including certifications, increased training, and professional development opportunities for principals (Herman et al., 2017). An example of such a program is the McREL Balanced Leadership Program, which consists of 10 2-day training sessions focusing on 21 identified leadership responsibilities and has been shown to have a beneficial effect on principal self-efficacy with respect to leadership practices as well as principal and teacher turnover (Jacob et al., 2015). This top-down approach of intervening in underperforming schools at the level of school leadership has been shown to have an overall positive effect on school culture while being more efficient than interventions focused on individual teachers, as principals have a wider reach and can disseminate information to the rest of the school staff (Herman et al., 2017).

ESSA requires school leaders to focus on three core principles in developing a plan for school improvement: a focus on meaningful learning, development of professional capacity, and targeted allocation of resources. Together, these should be approached through a mindset of continuous improvement (Darling-Hammond et al., 2016). In the state of New Jersey, continuous improvement is explicitly considered via a requirement for yearly progress monitoring as a part of the annual needs assessment and planning cycle for schools identified as underperforming and receiving funds to be put towards improvement. This involves setting timelines for monitoring progress towards specific, actionable goals and regular review of data to assess the efficacy of interventions (Riley et al., 2019).

A case study focusing on strategies employed by an underprivileged, underperforming high school in New Jersey as a part of the school improvement process identified building a team and cultivating trust as the most significant leadership strategies that should be adopted by other principals involved in this process. The study additionally identified increased reliance on datadriven approaches, improved professional development opportunities, and willingness to listen to feedback as important skills that can be employed by principals as a part of the school improvement process (Shelton, 2020). Increased utilization of stakeholder feedback was also identified as an important component of the school improvement process in a study focusing on principals at public elementary schools throughout the northeastern United States (Harrison, 2021). This willingness to listen is consistent with implementation of distributed or collective leadership strategies and has been identified as having a beneficial effect on collective efficacy (Ni et al., 2018). Distributed leadership has also been recognized by the Consortium for Policy Research in Education as a useful strategy for achieving meaningful and sustainable school improvement (Supovitz et al., 2019). A separate case study focusing on the experiences of teachers involved in the New Jersey school improvement process identified increased participation in professional development opportunities as the primary way in which teachers were involved, and this was associated with an increase in teachers' sense of self-efficacy (Mount, 2018).

Difficulties Faced by Urban Schools

There is a persistent achievement gap affecting the academic performance of poor and minority students relative to their peers. Schools in urban areas are more likely to serve these populations as well as to be overcrowded and underfunded (Kainz, 2019; Knight, 2017). This

population has also been identified as having the greatest potential for improvement with Title 1 interventions (Kainz, 2019). It is thus all the more important that these schools continue to receive Title 1 funding, which is jeopardized by their being identified as underperforming.

In addition to challenges related to funding and student demographics, urban schools are often located in areas that are affected by higher rates of violent crime, high traffic and pollution levels, and generally less safe environments, all of which have been shown to impact absenteeism and school performance negatively (Berman et al., 2018). Several factors that can adversely affect school scores occur at a disproportionately high level in schools in urban areas, including rising poverty rates, homelessness, crime, abandonment of education, teen pregnancy, and weapons possession (Barnett & Stevenson, 2016).

Teachers in urban schools report high levels of student misbehavior and disengagement, with almost half of teachers reporting that they encounter aggressive behaviors from students on a weekly basis (Camacho & Parham, 2019). Schools serving disadvantaged students are thus often forced to improve on factors unrelated to education itself. This can prevent a challenge for urban teachers, who are often less experienced and have fewer resources to help them than teachers in suburban areas (Knight, 2017). Urban schools also face barriers to teacher recruitment and retention, which has been shown to have a negative impact on student performance (Wronowski, 2018). Overall, the stresses of working in an urban school serving high-poverty minority students, coupled with the added pressures of NCLB and the legacy thereof, can have a negative effect on teacher morale (Byrd-Blake et al., 2010). This is likely to impair self-efficacy among teachers, which can, in turn, have negative effects on student outcomes.

Student Self-Efficacy

Student self-efficacy is the belief of students that they are capable of succeeding in their education. Central to the research questions addressed by the present study is the premise that student self-efficacy is enhanced by teacher self-efficacy as well as collective efficacy. Student self-efficacy is associated with increased emotional and behavioral engagement and improved academic performance (Bandura, 1993; Olivier et al., 2018; Zimmerman, 1995; Zimmerman et al., 1992). Bandura proposed that self-efficacy incorporates cognitive, motivational, affective, and selection processes, which collectively influence an individual's mindset about their capacity to learn or otherwise achieve a certain goal (Bandura, 1993). Bandura alternatively described self-efficacy as being highly related to an individual's belief in their own agency, and in this way, self-efficacy can be related to Dweck's mindset theory, described below. Students with high self-efficacy are more motivated to learn and show greater persistence, consistent with a greater belief in their own capability and expectations of success (Bandura 1993; Zimmerman, 1995). Bandura (1993) also demonstrated that self-efficacy was a greater predictor of success in a mathematics task than was ability itself.

Importantly, whereas efficacy theory describes perceived efficacy as a purely innate drive, Bandura proposed in his self-efficacy theory that self-efficacy is an acquired trait (Zimmerman, 1995). While some students may be more naturally inclined towards higher levels of self-efficacy than others, several studies have demonstrated that self-efficacy can be deliberately enhanced either through cultivation by parents at an early age or later in life (Bandura, 1993; Shin & Bolkan, 2021; Uchida et al., 2018). Uchida et al. (2018) demonstrated that increased self-efficacy could be manipulated in a classroom setting by inducing successful performance. A cohort of 12- and 13-year-old students performed a cognitive task wherein one

group of students received significantly easier problems than the other and thus had higher success rates. Significantly, the group of students who underwent manipulated success reported higher levels of self-efficacy that persisted for as long as a year following the study (Uchida et al., 2018).

Self-efficacy and intrinsic motivation can also be enhanced by specific instructional strategies, such as providing students with sufficient intellectual stimulation as part of a transformational leadership approach to teaching (Shin & Bolkan, 2021). Effective teaching techniques included using an interactive teaching style, challenging students, and encouraging independent thought. Additionally, students' perceptions of the learning environment, and specifically their sense of shared control and whether they were encouraged to make contributions to the classroom, were predictive of behavioral engagement and self-efficacy (Sökmen, 2021). Alternatively, negative teacher behaviors have been shown to have a detrimental effect on self-efficacy, such that students who were admonished by teachers and perceived them as being dissatisfied had significantly lower levels of self-efficacy (Shukla et al., 2020).

Teacher Self-Efficacy

As mentioned above, increased teacher self-efficacy is positively associated with student self-efficacy, which, in turn, has a highly beneficial effect on student performance. Several factors have been identified that promote a sense of self-efficacy among teachers, including a perception of a safe and supportive school environment (Reaves & Cozzens, 2018). The same study identified teacher self-efficacy as being promoted by increased supervision from administrators and clarity of expectations, and additionally associated with increased motivation

and better relationships with other staff members. Teacher self-efficacy can be directly related to instructional quality, which is, in turn, related to student motivational beliefs (Burić & Kim, 2020). Increased teacher confidence in their abilities has also been associated with improved relationships with students and decreased levels of conflict (Hajovsky et al., 2020). In turn, teacher self-efficacy has also been shown to indirectly mitigate the negative effects of bullying and victimization on student self-esteem via improved social dynamics and students' perceptions of the student-teacher relationship (van Aalst et al., 2021). Teachers with higher levels of self-efficacy are also more likely to engage in mastery-oriented teaching practices focused on creativity, understanding, and meaningfulness, as compared to teachers with lower levels of self-efficacy who are more likely to employ performance-oriented teaching strategies (Poulou et al., 2019).

In addition to its positive impact on students, teacher self-efficacy has been shown to be associated with decreased burnout and lower stress levels for teachers themselves. This was particularly true in urban schools, in which teachers are more likely to report stress and burnout (Bottiani et al., 2019). In fact, a meta-analysis of data from the Teaching and Learning International Survey, including results from over 400,000 teachers around the world, indicated that teacher self-efficacy was highly associated with increased job satisfaction (Kasalak & Dagyar, 2020). Teacher self-efficacy is not a universal quality and can be divided into, among other categories, self-efficacy towards classroom instruction and self-efficacy towards student engagement. Inadequate self-efficacy in only one of these domains has been shown in some circumstances to reduce teacher job satisfaction, while interventions targeted to increase teacher self-efficacy must consider not only general confidence but also self-efficacy within a given domain (Perera et al., 2019).

Professional Development

Professional development for teachers has been shown to increase feelings of competence and self-efficacy, with important implications for student outcomes (Schipper et al., 2018).

Teacher professional development may be subject-specific or focus on broader educational practices such as classroom management, with individual teachers having different needs (Perera et al., 2019). For example, professional development training targeted specifically towards mathematics teachers can have a beneficial effect on self-efficacy related to both subject knowledge and general classroom management practices (Carney et al., 2016; Ross & Bruce, 2007). Professional development opportunities have also been developed to improve teacher self-efficacy particularly with respect to the use of the internet and other technologies in the classroom (Watson, 2006) as well as for specific instructional techniques such as inquiry-based science teaching (Tsivitanidou et al., 2018). Professional development training can also take multiple formats, with one study indicating that formats that include follow-up coaching and focus on mastery have the strongest positive effects on teacher self-efficacy (Tschannen-Moran & McMaster, 2009).

School Collective Efficacy

The overall concept of collective efficacy derives from SCT and is defined as a collective belief in the ability to work together to achieve a common goal. Bandura (1997) noted that "perceived collective efficacy is an emergent group-level attribute rather than simply the sum of members' perceived personal efficacies" (p. 478). As such, while individual self-efficacy is related to collective efficacy, it is not a sufficient replacement for a shared belief in the group's effectiveness as a team. Lee and Louis (2019) suggested that collaboration, trust, and social control are key behavior-oriented elements of group effectiveness. Collaboration and trust are

two forms of social capital, instilling group members with a common community consciousness and the perception that common goals are mutually beneficial.

In a school setting, teacher self-efficacy and effective school leadership practices are predictive of perceived collective efficacy (Cansoy & Parlar, 2018). Furthermore, collective efficacy among teachers is associated with increased teacher leadership and communication as well as a shared goal of academic excellence. Additionally, collective efficacy is tied to increased job satisfaction, increased commitment to their job, and more positive attitudes towards professional development (Donohoo, 2018). Consistent with an increased openness to professional development opportunities, effective implementation of professional learning communities is positively associated with collective efficacy, as well as increased collaboration between teachers and higher levels of student achievement (Voelkel & Chrispeels, 2017).

School collective efficacy has, in fact, been widely established to have a positive impact on student performance (Bandura, 1993; Donohoo, 2018). Not only is collective efficacy beneficial to student performance in general, but it has been specifically demonstrated to reduce the achievement gap between Black and White students (Goddard, 2017). Collective efficacy also promotes parent and community engagement, contributing to improved relationships not only between staff but also with other members of the learning community (Kirby & DiPaola, 2011).

Principal and School Collective Efficacy

Leadership plays an important role in determining organizational culture and maintaining a commitment to the collective goal. Leaders not only play a critical role in developing and implementing policies to direct an organization towards a goal, but also serve as role models for those with whom they work. Transformational leadership, in particular, can have a positive

impact on an organization's culture, as these leadership qualities are associated with vision, motivation, and innovation (Bass & Avolio, 1993). This applies not only to corporate institutions but also to educational settings.

Attitudes of principals have been shown to have a significant impact on the overall school culture, and use of transformational leadership strategies by principals has been shown to increase school collective efficacy (Ninković & Knežević, 2018). Transformational leadership by principals was also associated with increased teacher job satisfaction and commitment to the institution. Some of the transformational leadership practices employed by principals in this study included providing individualized support and intellectual stimulation, building programs to encourage collaboration and parent engagement, and contributing to the development of a shared vision (Ninković & Knežević, 2018). Additionally, teacher self-efficacy and collective efficacy were reciprocally influential with principal self-efficacy. A separate study identified principal leadership strategies as having direct and indirect effects on teacher professional learning; indirect effects were at least, in part, mediated by an increase in teacher self-efficacy (Liu & Hallinger, 2018). In a case study of a high-poverty rural school, principal self-efficacy was associated with increased collective efficacy and promoted effectiveness of the school by improving teaching and learning and enhancing the school's overall climate. Teachers identified maintaining an instructional focus, developing teacher leaders, and leading by example as critical behaviors that this principal used to foster collective efficacy (Versland & Erickson, 2017).

Another means by which principals can foster collective efficacy is by encouraging collective leadership. Ni et al. (2018) found that increased influence over decisions by other stakeholders did not diminish the decision influence of principals and has the potential to create a more positive and inclusive school environment. Sun and Xia (2018) demonstrated that such

distributed leadership strategies were associated with increased teacher job satisfaction which was in part mediated by an increase in teacher self-efficacy. Liu et al. (2021) expanded on this finding, demonstrating that the beneficial effect of distributed leadership on teacher self-efficacy was, in part, mediated by collaboration and a supportive school culture. Thus, it has been widely demonstrated that principals can exert substantial influence over school culture and collective efficacy, and specific leadership strategies such as transformational leadership can help facilitate this. Distributed leadership has also been suggested by the Consortium for Policy Research in Education as an effective technique to bring about meaningful and sustainable school improvement (Supovitz et al., 2019).

Theoretical Framework

Social Cognitive Theory

Bandura's Social Cognitive Theory (SCT) contended that self-efficacy is the driving force that drives people to achieve more than originally they thought of their ability to achieve (Donohoo & Katz, 2017). The theoretical framework of this present study also asserted that people with a high level of self-efficacy tend to recognize themselves as accomplishing tasks that they interpret as difficult. Bandura found that the higher the level of self-efficacy, the more resilient, sustainable, and determined they were to achieve successful results. Self-efficacy is defined as a social cognition of students' belief in their ability to be successful in new learning. It means recognizing that everyone has the unique ability to learn, organize, and apply new skills. The SCT states that social interaction in the learning environment is one of the most influential factors in student learning. According to Bandura (1993), cognitive processes, motives, and effects are interrelated and mixed and act as catalysts for interactions. Academic performance refers to how students and teachers perceive their ability to succeed as well as the capacity of the

people with whom they interact. Teachers with a high level of self-efficacy can motivate and engage students in the learning process.

Mindset Theory

Mindset theory, developed by Carol Dweck (2006), describes the phenomenon whereby individuals who believe that certain skills and traits can be developed as opposed to being innate are more likely to find success in these endeavors. This is referred to as a growth mindset, whereas the alternative belief that most individuals are born with certain abilities which cannot be changed is referred to as a fixed mindset (Dweck, 2006). In short, mindset theory demonstrates that an individual's self-perception can profoundly affect the way they live their life. This is particularly true with respect to beliefs about that which an individual is capable of as well as attitudes towards their malleability. A growth mindset is not necessarily the same as a positive outlook, but it is simply the belief that one can improve oneself with targeted effort (Dweck, 2016). The framework has been applied to several different fields and has implications for social and emotional skills development, physical and technical skills, and perhaps most commonly—as is discussed here—education (Bernecker & Job, 2019; Dweck, 2006).

Mindset theory was originally developed in response to the observation that different children respond very differently to obstacles. While some children were easily deterred by difficult problems, others actively sought these out, seemingly enjoying the challenge (Bernecker & Job, 2019). Consistent with this, individuals with a growth mindset tend to value effort put into a task as opposed to a mere outcome. Alternatively, this might be thought of as an emphasis on learning goals instead of performance goals (Bernecker & Job, 2019). In fostering a growth mindset, Dweck recommended praising children for their hard work and efforts rather than for their innate abilities and achievements (Dweck, 2006). Mindset interventions in adults have been

shown to increase challenge-approach motivation and self-determination and to be associated with an internal locus of control (Burgoyne et al., 2018).

Importantly, a growth mindset has not only been shown to improve student achievement, but it also can itself be taught (Yeager & Dweck, 2012). Among the first to demonstrate this using a controlled study, Blackwell et al. (2007) performed an intervention study in which seventh grade students attended an 8-week workshop consisting of 25-minute sessions once per week focused on the malleability of intelligence. Following the intervention, students in the experimental group demonstrated increased classroom motivation and a math grade point average 0.3 points higher than that of a control group who attended workshops that included the same information about the brain and study skills but not about malleable intelligence (Blackwell et al., 2007). Through a longitudinal study, the authors also found that students' preexisting belief in incremental theory, or tendencies towards a growth mindset, was predictive of future academic performance (Blackwell et al., 2007). Despite extensive findings now demonstrating that educating students on resilience and promoting a growth mindset result in improved academic and social outcomes, this remains an area for increased focus in most schools today (Yeager & Dweck, 2012).

Gaps in the Literature

The literature on self-efficacy and collective efficacy clearly demonstrates that these phenomena are mutually dependent and have a reciprocal influence. Furthermore, in an educational context, collective efficacy contributes to improved student outcomes and can be fostered by effective leadership strategies. Teacher collective efficacy has been shown to have a positive effect on the school environment for all parties involved in the educational process. However, while collective efficacy is generally recognized to have a positive impact on both

school culture and academic outcomes, there has been little focus on how this relates to the school improvement process for schools identified as underperforming or how this may be influenced by an urban environment. Specifically, there is little research concerning how urban principals perceive of, respond to, and facilitate collective efficacy in this context. This present study explored the perceptions of principals about collective efficacy and how these perceptions influenced their leadership strategies related to the school improvement process. Finally, this study related findings about principals' perceptions of collective efficacy to the theoretical frameworks of social cognitive theory and mindset theory, both of which have been used to describe phenomena associated with self and collective efficacy.

Summary

This chapter presented a comprehensive literature review, providing context for the present study as it relates to the role of leadership practices in promoting collective efficacy in urban schools. The legislative history of student achievement gaps and how this has led to labeling schools as failing to meet standards, as well as the challenges associated with urban school districts, are presented, as is a review of self-efficacy and collective efficacy in the context of various educational roles. Finally, the theoretical frameworks of social cognitive theory and mindset theory were described in detail and related to the central research questions.

This literature review established a firm basis upon which to conduct the present research study, which interrogated principals' perceptions of collective efficacy in urban K-8 schools designated for the school improvement process in the state of New Jersey. Chapter 3 next builds on the framework established here, elaborating on the specific research questions addressed by this study and outlining the planned research objectives and methodology. It also includes a discussion of relevant ethical considerations, assurances of reliability and effectiveness of the

survey instruments, and limitations of the study, as well as a plan for how the present study addressed these issues.

CHAPTER 3: METHODOLOGY

Introduction

This chapter describes the methods of carrying out research to answer the research questions, as well as the population and samples that were selected for the study. In addition, this chapter provides a synopsis of the participants who engaged in the study and the setting in which the study took place. A description of the devices used in the studies, data acquisition process and data analysis and reporting procedures are also reported. The interview process and protocol are discussed, followed by a detailed summary of the method for data collection. The method for data analysis is also described, followed by a conclusion and discussion of proper considerations for the research study.

Purpose Statement and Research Questions

The purpose of this qualitative study was to explore principals' perceptions of collective efficacy in urban New Jersey schools designated for the school improvement process. Schools that are publicly flagged and identified as underperforming may have exhausted effective structural and collective efficacy strategies within the school community based on current conditions. As a result of school improvement identifiers, one might assume that school district officials, teachers, students, and families internalize public perception and enact minimal collaborative practices. This fixed mindset can inadvertently hinder the implementation of practices that promote sustainable progress and collective efficacy practices. Previous studies have confirmed a relationship between student self-efficacy and academic performance (Cansoy & Parlar, 2018). With the present study, the researcher sought to better understand the principals' perceptions of collective efficacy in underperforming schools in New Jersey as a means of

school improvement. As stated in Chapters 1 and 2, the purpose of this qualitative study was to explore principals' perceptions about collective efficacy in urban K-8 New Jersey schools designated for the school improvement process. The research conducted was guided by the following questions and sub-questions:

RQ1: What are principals' perceptions of community collective efficacy as a means towards school improvement?

SQ1a: How do principals describe the value/possibility of community collective efficacy?

SQ1b: How do principals describe the challenges of community collective efficacy?

Research Design

The researcher selected a qualitative study design to acquire emotion, insight, and stories as told from the perspective of educators in schools designated for improvement. The use of a qualitative research design can be thought of as a rough sketch to be filled in by the researcher as the study proceeds (Frankel & Devers, 2000). This research was guided by a qualitative methodology and a narrative research design. For this research, narrative analysis was a selected method used to capture the lived experiences of principals and their perceptions of collective efficacy. Narrative research was used to describe the principals' experiences through their own stories (Butina, 2015). To conduct the research for this study, the researcher used a semi-structured interview approach with the principals. The study was designed to understand the structures that have been effective in underperforming schools and to determine how collective efficacy assists leaders in improving the schools' status.

Qualitative methods are appropriate when the goal of a study is to better understand why or how a phenomenon occurs, rather than quantifying the prevalence or extent of a phenomenon (Taylor et al., 2015). Qualitative research also involves the collection of detail-rich data. In the case of this research, the aim was to explore the principals' perceptions of collective efficacy in urban K-8 schools in New Jersey designated for the school improvement process and whether these perceptions play a role in their decision-making. Principals may seek out and implement specific methods for their schools' improvement process based on their perceptions of student performance within the school community.

Data Collection

The researcher obtained direct permission to recruit participants via district email from the deputy superintendent and superintendent of the Central Public School district. To this end, the researcher sent out email invitations introducing herself and explaining the study. The solicitation was sent out at the completion of the IRB approval process. This research study's data sources included an interview survey with eight open-ended interview questions designed to obtain answers to the research questions (please see the semi-structured interview section below). These open-ended interview questions, which took approximately 30 minutes to complete, were emailed to 15 potential participants who responded to the researcher's call for participation. Of those 15 participants, only 10 met all the inclusion criteria. Briefly, one respondent was not a principal; three respondents were not principals in the state of New Jersey; and one respondent was a director, not a principal, of an elementary school in Trenton, New Jersey. These five participants were disqualified from participating in the study.

Upon confirmation that potential participants met the inclusion criteria, the researcher distributed informed consent forms that were required to participate in the study. Upon receipt of the completed informed consent forms, the researcher distributed the open-ended interview surveys to the participants via email. Participants were able to complete the interviews on their own time, in the location of their choosing. Each participant emailed the completed survey to the researcher. Although the interview data was already in written form directly from the participants, member checking was still conducted (Candela, 2019). There was a need for member checking since member checking is a method used to address the study's credibility (Johnson et al., 2019). The participants were assigned pseudonyms, namely P1, P2, ..., and P10, to protect their identity and ensure anonymity. The surveys were directly uploaded without alteration to the NVivo Software for analysis and coded by themes to determine the similarity, dissimilarity, or collaboration of the participants' responses.

Research Protocol

The research protocol for data analysis was conducted as described in this chapter. The data analysis procedure consisted of the following steps. The researcher first performed four critical readings of each interview survey. In the first reading, the researcher read each survey for overall impressions. In the second reading, the researcher read each survey line-by-line to become familiar with all of the data. Third, the researcher read the surveys by each interview question instead of by participant. This allowed the researcher to identify important themes that emerged from each survey question. It also provided the researcher with insight into which survey questions the participants "grouped" into common themes. For example, Survey Questions 1 and 3 dealt with the participants' perceptions of collective efficacy generally, and

then specifically the state of collective efficacy at their schools. These questions were grouped together to comprise Subsection 1 of the data analysis and results. Similarly, Survey Questions 2 and 4 were grouped together to form Subsection Two, which analyzed the effect of the *needs improvement* school rating on collective efficacy. Survey Questions 5 through 8 dealt with how the principals utilized collective efficacy in various leadership-based activities; these survey questions were grouped to form Subsection Three of the data analysis and results section. Finally, the researcher did a fourth reading of the surveys, by participant, to regain a holistic appreciation of the entire data set.

The researcher next reviewed the interview transcripts to determine if the data correlated to the research question. A thematic coding system was utilized to code the data for each survey question. Within each research question, codes were given to different ideas that the participants expressed. For example, Interview Question 2 asked the participants to explain the impact, if any, the label *school designated for improvement* has had on their school community. The themes for this question were derived from exhaustive cases—namely, the participants could report the rating as having a *positive* effect, a *negative* effect, or *no effect*. Within each of these three themes (*positive*, *negative*, *no effect*), subthemes were found by analyzing participants' responses. For example, for this question, within the positive theme, a subtheme of having a *common goal* emerged from the participants' responses. All survey questions were coded in this manner. The codebook in Appendix A contains a comprehensive account of all themes, subthemes, and codes assigned to the participants' responses.

After the identification of themes and different ideas contributing to the development of each theme, the researcher chose direct quotations from participants to help explain the themes.

If a variety of participants had similar responses within a theme, the researcher ensured that the chosen quotation within each data table was representative of the general sentiment of all participants. In this study, there was little disagreement between participants in that many of them expressed the same thoughts or individualized thoughts that were not replicated by any other participant. However, participants did offer contrasting positions on a few subthemes.

Notably, any and all differential opinions and contrasts are noted in the data analysis section.

This type of data was not chosen by the researcher for omission. Finally, after the researcher identified themes, subthemes, and quotations, the researcher did a fifth reading of the entire data set to ensure that all themes, subthemes, and quotations chosen were representative of the participants' intentions and perceptions.

Participants

All of the participants in the study were principals at public elementary and high schools in the state of New Jersey. Further, all of the participants led schools that were placed on the state's *school in need of improvement* list. The researcher attempted to recruit participants statewide in New Jersey. However, all participants who responded to the call for participation led schools in a large urban city of New Jersey. As such, the target population of the study was modified to be principals who led K-12 schools in need of improvement in this large urban city. This modification may limit the generalizability of the study, a concept that is discussed further in Chapter 5 of this dissertation. The demographics of the participants are presented in Table 1.

Table 1Demographic Characteristics of the Participants

| Participant | Years of Experience as a Principal | Current School | Number of Years as Principal at Current School |
|-------------|------------------------------------|-------------------|--|
| P1 | 5 | Elementary | 3 |
| P2 | 1 | Elementary | 1 |
| P3 | 8 | Elementary | 6 |
| P4 | 3 | High School | 5 |
| P5 | 2 | Elementary | 1 |
| P6 | 2 | High School | 1 |
| P7 | 3 | Elementary | 2 |
| P8 | 4 | High School | 1 |
| P9 | 2 | Elementary | 5 |
| P10 | 6 | Elementary | 6 |

In this large urban city in New Jersey, the public school system only consists of two types of schools—namely, elementary schools (K-8) and high schools (Grades 9-12). In the study, 70% of the participants were elementary school principals, while 30% of the participants were high school principals. The principals had, on average, 3.6 ± 2.1 years of experience of being a principal at any school and 3.1 ± 2.1 years of experience of being a principal at their current school. Interestingly, these data indicated that the majority of the principals have spent their principalship at the same school in which they are currently serving.

Semi-Structured Surveys

The researcher created an interview survey to ask participants open-ended questions regarding their perceptions of collective efficacy at their respective schools. The survey questions prepared were clear, concise, and well-defined to obtain quality responses from

participants. The interview questions and how they correlated to each research question, subresearch questions, and theoretical framework are shown below.

- 1. How would you define collective efficacy? [RQ1, Social Cognitive Theory]
- 2. Explain the impact if any the label *school designated for improvement* has had on your community. [SQ1a, SQ1b, Theory of Mindset]
- Describe the state of collective efficacy in your school. [RQ1, Social Cognitive Theory]
- 4. Can you share how your staff and students feel about being employed and attending a school with a lower performance rating? [RQ1, SQ1a, SQ1b, Theory of Mindset]
- Describe the process you engage in to develop strategic plans and strategic decisionmaking at your school. Explain the role of collective efficacy in the process. [SQ1a, Social Cognitive Theory]
- 6. How does/did your perceptions about collective efficacy influence your decisions for the school during the improvement process? [RQ1, SQ1a, SQ1b, Theory of Mindset]
- 7. During this process, did you make or have you made purposeful decisions based on your perceptions about collective efficacy? Why or why not? [RQ1, Social Cognitive Theory]
- 8. Explain the process for identifying interventions for your targeted population. Do your beliefs and perceptions about collective efficacy influence your decision-making with regard to interventions? [RQ1, SQ1b, Theory of Mindset]

Saturation

The sample size of qualitative research varies according to the amount of data that must be obtained pending data saturation (Guest et al., 2020). Attaining data saturation is a cyclical process. The researcher identified themes throughout the data analysis and kept building on the themes as additional participants were interviewed until they no longer yielded different themes (Hennink & Kaiser, 2021). The sample size of 10 participants was selected because this sample size generally ensures that data saturation is reached in interview-based qualitative studies (Guest et al., 2020; Vasieleiou et al., 2018). Specifically, Guest et al. (2020) demonstrated that a sample size of 10 participants led to data saturation in 92% of interview-based qualitative studies.

Accordingly, the present researcher observed that data saturation was generally achieved after conducting the 10 interviews, as the same ideas were expressed in the 10 different interviews. This led the researcher to conclude that data saturation had been reached and no further interviews would lead to the identification of different ideas or themes.

Reliability and Validity

Reliability denotes the consistency of an instrument (Creswell & Creswell, 2018, p. 153). To ensure reliability, all participants were asked the same interview questions developed from and aligned with the same research questions using an interview protocol; they were also provided with the same amount of time during the interviews. The interview questions and the interview protocol were shared with an expert panel to elicit feedback on the clarity of the questions and whether they would provide information to address the research questions. The feedback was then incorporated into the interview protocol. The school principals vetted to participate were not a part of any group who made up the participants for the study. The researcher did not use an independent transcriptionist that could potentially compromise the

confidentiality of the research study. Instead, the researcher used NVivo to transcribe the responses to the interviews. Finally, the researcher made a concerted effort to adhere to all the conditions set forth by the Institutional Review Board (IRB). As Creswell and Creswell (2018) said, "Validity is one of the strengths of qualitative research and is based on determining whether the findings are accurate from the standpoint of the researcher, the participant, or the readers of an account" (p. 199). To ensure validity, the researcher recorded the interviews using pseudonyms (i.e., P1, P2), and the transcripts of these interviews were sent back to the participants for member checking. Lincoln and Guba (1990) related the use of member checking in qualitative research as the "most crucial technique for establishing credibility" (p. 314). This process gave the participants an opportunity to review their responses to ensure that their words were transcribed accurately. Any changes found by participants were changed immediately to best capture their point of view.

Limitations

One limitation of the study is that it was conducted solely within the Central Public School district. The researcher intended for participants to be drawn from multiple counties in New Jersey. However, the only participants who volunteered for the study who met the inclusion criteria were from Central Public Schools. Therefore, the actual target population of the study is Central Public School principals who lead schools in need of improvement. This may impact the transferability of the study to other counties in New Jersey; furthermore, the findings of the study may not be generalizable to other schools in other states. However, the researcher believes that the results of the study are transferable to any urban school district in the United States, as research as shown that most urban schools experience the same challenges and barriers (Howard & Milner, 2021).

Another limitation of the study was the sample size. The researcher was only able to purposefully recruit 10 participants for the study. Low numbers of participants can lead to a lack of data saturation. However, in this case, the researcher believed that data saturation was reached. In some cases, 80-90% of the participants highlighted ideas within the same theme (see the codebook in Appendix A).

A third limitation of the study is that the survey was done via email rather than using inperson interviews. The researcher was not present when the participants were completing the
surveys. Therefore, the researcher was unable to read and interpret the participants' nonverbal
responses, including body language. As such, it is impossible for the researcher to make
suppositions based on the participants' nonverbal communication.

Finally, the paper-based survey also served as another important limitation. By confining the interview to a specific set of questions on paper, the researcher was unable to ask follow-up questions when needed. However, this limitation was, in part, mitigated by the fact that the participants took care in explaining their thoughts and did so thoroughly.

Summary

Chapter 3 described the use of the narrative qualitative approach needed for this study. This chapter also described the participants surveyed, the measuring instruments used for data collection, and the data analysis procedure. Also discussed were the effectiveness and reliability of the survey, important ethical considerations related to the survey, and limitations.

CHAPTER 4: RESULTS

Introduction

Collective efficacy is "the group's shared belief in its conjoint capabilities to organize and execute courses of action required to produce given levels of attainments" (Bandura, 1997, p. 477). While important in all organizations, collective efficacy is significant in educational settings (Goddard et al., 2004). Within education, collective efficacy can transform school settings into vibrant communities where collaborative learning is facilitated by school administrators, support staff, teachers, and the students themselves (Goddard et al., 2004). Consequently, schools needing improvement could turn to collective efficacy to help improve their school communities, thereby increasing student achievement. The school leaders who participated in this study utilized inclusive structures. These structures included professional development that focused on tiny shifts academically by using PLC cycles to analyze data. Teaching sprints were introduced to school administrators and instructional coaches by the Office of Teaching and Learning as a means of analyzing small data output connected to the curriculum. There was evidence in six of the 10 schools that principals and teachers were engaging in teaching sprints that allowed all stakeholders to come together to determine a focus for practice improvement. However, it is presently unknown how principals who lead schools that need improvement view collective efficacy and its role in improving the school community.

In Chapter 4, the researcher presents the data analysis and results from this study on principals' perceptions of collective efficacy. The data analysis and results are divided into three subsections based on the research questions to address an important gap in the literature. The following research questions and sub-questions were developed to guide this study:

RQ1: What are principals' perceptions of community collective efficacy as a means towards school improvement?

SQ1a: How do principals describe the value/possibility of community collective efficacy?

SQ1b: How do principals describe the challenges of community collective efficacy?

Data Analysis and Results

The data analysis and results sections are divided into three subsections based on the research questions. Participants answered questions related to their definition of collective efficacy, the state of their school collective efficacy, how collective efficacy affects decision-making, identifying interventions, and the role of collective efficacy in strategic management. Participants also filled out a questionnaire on the school improvement process. The data were organized into tables based on the answers to the questionnaire; however, the researcher organized the data results here in subsections based on the research questions. In the first subsection, the researcher presents the principals' definitions of their perceptions of collective efficacy (Table 2), their perceptions of the state of collective efficacy at their schools (Table 3), and their perceptions of decision-making (Table 4). In the second subsection, the principals' perceptions of the benefits of collective efficacy are examined, including positive effects on the school and decision-making. Finally, in the third subsection, the challenges associated with collective efficacy were examined, including mind shift, works in progress, identifying interventions (Table 4), decision-making (Tables 6-7), and improvement ratings (Table 8).

Subsection One: Principals' Perceptions of Community Collective Efficacy as a Means of School Improvement

Principals' Definitions of Collective Efficacy

Before beginning to understand the principals' perceptions of community collective efficacy as a means of school improvement, it was necessary to understand their definitions of collective efficacy. Therefore, the first survey question asked the principals for their definition of collective efficacy. This question resulted in three general definitions; some participants defined collective efficacy as an ability, others defined it as collaboration and teamwork, and the remainder of the participants defined it as working towards a common goal (Table 2).

 Table 2

 Principals' Definitions of Collective Efficacy

| Definition of Collective Efficacy | | | |
|-----------------------------------|------------------------|--|--|
| Theme | Participants | Excerpt | |
| Ability | P3, P5, P6, P8, P10 | "A group's shared understanding gives that group the ability to control other behaviors. It's a way to create an order using a system of understandings" (P10). | |
| Collaboration, Teamwork | P2, P4, P5, P6, P9 | "Collective efficacy is the belief that everyone can and will succeed" (P9). | |
| Works Towards a Positive Goal | P1, P2, P5, P7 | "Having a sense of togetherness as an organization. Using that togetherness to foster common beliefs in the ability to accomplish goals and to work effectively" (P5). | |

Collective Efficacy as an Ability. Three general themes emerged from the principals' definitions of collective efficacy. Some defined collective efficacies as an ability that is shared by a group. For example, P3 said that collective efficacy is "a group's ability to produce the desired result." Similarly, P8 said, "I would define collective efficacy as the ability to have

collective trust. [It is] confidence individuals within an organization have towards achieving success." In this way, some of the participants believed that collective efficacy was a matter of ability, in that the organization needed to have the ability or capacity to achieve collective goals. It should be noted that ability is an intrinsic property of a group or an organization (Riegelsberger et al., 2005). Ability can be nurtured and fostered to meet the goals of an organization (Lysova et al., 2019), but, ultimately, the ability to effect change begins with intrinsic motivation and the properties of individuals within a group (Locke & Schattke, 2019). Therefore, bringing individuals in or out of the organization could directly affect the perception of the collective efficacy of the organization (Chen et al., 2019). Therefore, this suggests that strong positive leadership from principals and administrators can positively influence the perception of collective efficacy of schools.

Collective Efficacy as Collaboration and Teamwork. The majority of the principals surveyed in the study gave a definition of collective efficacy that focused on collaboration and teamwork. For example, P6 described:

I would describe collective efficacy as the ability to collaborate with parents, students, and staff (instructional and non-instructional) to hire, develop and sustain staff members who are firmly committed to transforming any underperforming academic institution.

This process requires ongoing communication, parental support, and ongoing monitoring.

(P6)

P6 highlighted the notion that collective efficacy is grounded in teamwork and collaboration between all parties and stakeholders, including staff, parents, and students. In the eyes of P6, collective efficacy is an ongoing process that requires the consistent cooperation of all parties. These sentiments were echoed by a variety of other participants. Indeed, P4 said:

Collective efficacy complements organizational systems, which are the underpinnings of collaboration.... Collective efficacy ensures that stakeholders work with fidelity to achieve school-wide goals to leverage the greatest impact to supersede pockets of the greatest throughout the school community. (P4)

Both P4 and P6 emphasized that collective efficacy requires collaboration between all stakeholders to achieve school-wide goals. Many other principals shared the same thoughts. For an exhaustive list of the principals' definitions of collective efficacy, please see Table A-1 in Appendix A.

Collective Efficacy as Work Towards a Positive Goal. The last theme that emerged from the analysis of the principals' definitions of collective efficacy was the notion that collective efficacy is working towards a positive goal. In many ways, this theme was intertwined with other themes, but was distinct in that work towards a positive goal implies there is a perception of action by individuals. For example, P5 intertwined the theme of collaboration, ability, and work towards a common goal. P5 said, "[Collective efficacy] is having a sense of togetherness as an organization. Using that togetherness to foster common beliefs in the ability to accomplish goals and to work effectively." Importantly, P5 highlighted that ability and working towards a goal are different. Ability is an intrinsic characteristic of an individual that can be fostered, whereas working towards a goal involves consistent action. Similarly, P1 argued that "Collective efficacy is the work that happens toward a shared and collective belief that all students can learn and excel." In this way, many of the participants believed that significant action and work are required to foster collective efficacy. This idea may seem incongruent with the idea that perceived collective efficacy is an ability of a group. However, abilities can be fostered, which can, at times, take a significant amount of work. In summary, the principals

perceived collective efficacy as an ability of a group to foster collaboration and teamwork in an effort to work together to meet a common goal.

Current State of School Collective Efficacy

Next, the principals were asked about the current state of their school's collective efficacy. This question resulted in a wide variety of responses. One principal (P5) reported their school had high collective efficacy; another principal (P3) reported that collective efficacy was initially positive and then turned negative; yet another principal (P10) reported that their school had low collective efficacy. Other themes mentioned by participants included: (a) the school had to shift its mindset; (b) improved collective efficacy led to positive effects in the schools; and (c) the school's collective efficacy was a work in progress. These results are summarized in Table 3. The *Resulted in Positive Effect* theme is examined in Subsection Two, and the *Mindset Shift* and *Work in Progress* themes are examined in Subsection Three.

High Collective Efficacy. One principal (P5) reported that their school had high collective efficacy, in part because the principal was adding another school to the school community. As such, this required a large amount of collective efficacy to join the schools together to function as one collective and effective community. P5 said:

My school community consists of two buildings that were recently merged for the first time. This upcoming school year requires laser-sharp focus in having the teachers and staff embark on a major shift in seeing the school community as one. (P5)

As such, P5 described that their school required a high collective efficacy to effectively join the two schools, even in spite of the *needs improvement* rating.

 Table 3

 Principals' Perceptions of the State of School Collective Efficacy

| State of School Collective Efficacy | | | | |
|--|-----------------------|--|--|--|
| Theme | Participants | Excerpt | | |
| High Collective Efficacy | P5 | "My school community consists of two buildings that were recently merged for the first time. This upcoming academic year requires laser-sharp focus in having the teachers and staff embark on a major shift in seeing the school community as one" (P5). | | |
| High Collective Efficacy, Then Low Collective Efficacy | P3 | "Initially, the collective efficacy is high due to the feeling of 'this is our school, and we must protect it.' Over long/extended periods of time, the momentum tends to wane" (P3). | | |
| Low Collective Efficacy | P10 | "I think it is lacking in my school. That is why we are currently dealing with chaos. There is a lack of structure and a lack of confidence in the administration. It feels like everyone is out for themselves" (P10). | | |
| Mindset Shift | P1, P9 | "There is a strong urge to be devoid of the 'school in need of improvement' label. Thus, we worked hard on mindset shifting" (P1). | | |
| Resulted in Positive Effect | P1, P2, P4, P7, P8 | "I am now in my third year at my school, and I am proud that although there were negative perceptions around our school's performance in my first year, we have worked hard to build trust again with all stakeholders While building trust and confidence in a school is iterative, we have made strides in building" (P8). | | |
| Work in Progress | P6, P8, P9 | "The state of collective efficacy at my school can be described as a work in progress where most stakeholders see the need to improve and support most of the changes implemented, but some tend to give up when immediate results are not evident or a challenge/ setback arises" (P6). | | |

High Collective Efficacy, Then Low Collective Efficacy. Like P5, P3 described an initial positive amount of collective efficacy. However, P3 observed that over time, the positive aspect of the schools' collective efficacy started to mellow, until there was little collective efficacy at the school. In the words of P3, "Initially, the collective efficacy is due to the feeling of 'this is our school, and we must protect it.' Over long or extended periods of time, the momentum tends to wane" (P3). P3 described a situation in which a school began an academic year with high collective efficacy, but it began to wane as the year progressed. This is perhaps not surprising, as teacher motivation and self-efficacy have been reported to decrease throughout the progression of the academic school year (Maulana et al., 2016). Similar trends have been seen with student motivation and self-efficacy throughout the academic year (Opdenakker et al., 2012). Consequently, a decrease in the self-efficacy of teachers and students would, in turn, lead to a decrease in collective efficacy (Goddard & Goddard, 2001; Skaalvik & Skaalvik, 2007).

Low Collective Efficacy. One principal reported that their school had a low collective efficacy this school year. P10 said, "I think [collective efficacy] is lacking in my school. That is why we are currently dealing with chaos. There is a lack of structure and a lack of confidence in the administration. It feels like everyone is out for themselves" (P10). P10 described a situation that was very different from situations with the other principals examined so far. For P10's school, the administrators and teachers seemed to be "out for themselves," meaning that they were more focused on self-preservation and job preservation than on the team, the collective. As such, this situation led to an instance where the school had low collective efficacy, when students do not rally behind their teachers and teachers do not rally behind their students. Indeed, in their study, Sørlie and Torsheim (2011) found that a lack of organizational structure led to an overall decline in the collective efficacy of the organization.

Utilizing Collective Efficacy for Identifying Interventions

The last survey question asked principals whether they utilized collective efficacy to identify interventions. Three themes were derived from the analysis of this question. Principals reported using a data-driven approach to identify interventions, a response-to-intervention (RTI) system, and finally, team-driven interventions (Table 4). Each theme is discussed in turn.

 Table 4

 Principals' Perceptions of Process for Identifying Interventions

| Process for Identifying Interventions | | | |
|---------------------------------------|---------------------------|--|--|
| Theme | Participants | Excerpt | |
| Data-Driven Process | P1, P6, P7, P9 | "We look at data to determine and isolate standards that students have not mastered, and we then group them and remediate those who have not" (P7). | |
| Response to Intervention (RTI) | P1, P2 | "We then determine what support will look like for those students. It could be various forms of RTI (Response to Intervention, Tutoring, Additional Guided Readings Sessions, Saturday Academy, etc." (P1). | |
| Team-Driven Intervention | P3, P4, P5, P6, P8, P9 | "Interventions are identified by a team that has direct knowledge/interaction with said population, ability to identify root causes, analysis of work samples or data, and can offer solutions for correction. The team is also responsible for benchmark review of the effectiveness of agreed upon intervention" (P3). | |

Data-Driven Process. Four principals (P1, P6, P7, and P9) reported utilizing a data-driven process for identifying interventions. Importantly, three of the four principals indicated that this data-driven approach was a team effort. For example, P1 said:

We determine which teachers are the strongest at which content. We analyze student data. Teachers create corrective instruction plans based on the data. This determines which students need support with which priority and/or pre-requisite standards. (P1)

P1 indicated that the intervention plans are a collaborative effort between teachers and administrative staff, which is fundamentally data-driven. P6 also emphasized a data-driven team process. P6 said, "The first step is pulling and analyzing student data as a team to determine which educational areas are in need of support for each child." This process was not unique to P1 and P6. Indeed, P7 also described a data-driven, team-oriented approach to interventions. P7 said, "We look at data to determine and isolate standards that students have not mastered, and we then group them and remediate those who have not [mastered the material]." Importantly, within this theme, each of the principals expressed using collective efficacy in a data-driven, team-oriented process to identify interventions.

Response-to-Intervention (RTI). Two principals (P1 and P2) reported using a response-to-intervention (RTI) system to help identify students in need of intervention. Importantly, both P1 and P2 described this process in a team-oriented way, suggesting that collective efficacy does play an important role in the identification of interventions. For example, P1 said, "We then determine what support will look like for those students. It could be various forms of RTI (response-to-intervention, tutoring, additional guided reading sessions, Saturday academy, etc.)." While P1's response may not seem interesting, the use of the word "we" is important, as it indicated that more than one person was involved in the intervention process. As such, even though P1 was utilizing a different method than other principals, they all still used collective efficacy to identify interventions. P2 echoed these sentiments, saying, "We use our RTI system and our I & RS system to make decisions collectively." Therefore, P2 also indicated that collective efficacy plays an important role in identifying interventions, even when using an RTI system.

Team-Driven Interventions. Six of the principals reported using team-driven interventions that directly involve collective efficacy. For example, P3 said:

Interventions are identified by a team that has direct knowledge/interaction with said population, the ability to identify root causes, analysis of work samples or data, and can offer solutions for correction. The team is also responsible for a benchmark review of the effectiveness of the agreed-upon intervention. (P3)

P3 described a system in which a team is utilized to identify students in need of intervention. The team also determines what intervention is needed and whether an intervention is effective. In this way, P3 described utilizing collective efficacy to identify interventions effectively. P4 similarly utilized collective efficacy in the intervention process. P4 said, "Identifying interventions for the targeted population is done in collaboration with central office leaders, content area specialists, research-based practices, and reflection." Thus, P4 also indicated that collective efficacy is important in the intervention process, given that administrators, as well as content area specialists, are tasked with identifying the correct intervention. Many of the principals reported similar processes, a complete list of which is detailed in Table A-8 in Appendix A.

Role of Collective Efficacy in Strategic Management

Strategic management is defined to be the process of setting goals, procedures, and objectives in order to make an organization more competitive (Jasper & Crossan, 2012). In particular, for schools, strategic management involves improving student achievement by substantially improving instructional practice and the effectiveness of teachers and principals by setting goals and making academic procedures and interventions (Greckhamer & Cilesiz, 2022). Each of the principals identified one of two themes as the role of collective efficacy in strategic management. The first of these themes is utilizing data analysis and SMART goals, and the

second theme is expanding the planning network to include more stakeholders. The results of this analysis are presented in Table 5.

 Table 5

 Principals' Perceptions of Role of Collective Efficacy in Strategy Planning

| Role of Collective Efficacy in Strategic Management | | | | |
|---|--|---|--|--|
| Theme | Participants | Excerpt | | |
| Data Analysis, SMART Goals | P1, P4, P6, P9 | "Being a school driven by DDI (Data-Driven Instruction), we always begin with Data Analysis to determine where we are & where we need to go. Teachers are part of this process. The Leadership Team develops SMART Goals within the Strategic Plan based on the Data. As we meet with teachers to support the creation of their SGO (Student Growth Objectives), they are clear about how their respective data fits into the scope of the whole" (P1). | | |
| Expand Planning Network | P2, P3, P4, P5, P6, P7, P8, P9, P10 | "Over time, I have come to expand the collective efficacy in hopes that the more hands/minds involved, the more widespread the breadth of work would become. I purposefully create multilayered committees that staff pick to focus on areas of interest to them. The hope is that they will hold each other accountable because they are genuinely interested and invested in that area" (P3). | | |

Data Analysis and SMART Goals. Four of the principals shared that they use data-driven analysis and SMART goals tools for strategic management in their schools. The SMART acronym was first used in business to guide the development of quality goals and objectives (Lazarus, 2004). With respect to goal making, SMART stands for specific, measurable, attainable, routine-based, and tied to a functional priority (Lazarus, 2004). P1 described using data-driven analysis coupled with SMART goals (Table 5). Importantly, P1 named this as part of collective efficacy, as the teachers and the leadership team are integral to the development of goals that fit the needs of the students as well as the needs of the school. P1 also described that

data analysis of student performance is integral to the development of the SMART goals (Table 5). P4 also addressed the use of data-driven analysis, saying, "Additionally, teachers are given two periods a week to work collaboratively to discuss student data, analysis and action plans to address deficiencies or maintain gains" (P4). In this way, P4, like P1, described that the teachers are integral to the establishment of goals and plans for strategic planning. As such, for these principals, collective efficacy plays an important role in the strategic management process with respect to goal setting and data-driven analysis.

Participants P6 and P9 described similar processes for data-driven analysis and goal setting. For example, P6 said, "We also pull and analyze student achievement data (including staff and student attendance) as well as teachers' observation data and resources available before setting goals for the strategic plan" (P6). Similarly, P9 said, "We prepared to develop strategic plans in Aug[ust] of every year. We brainstorm as a team and use previous data to set goals" (P9). Importantly, both P6 and P9 described their strategic management process as involving the individuals closest to student achievement—namely, the teachers. Working as a team, the principals and teachers design goals that are specific, measurable, attainable, and, most importantly, tied to the functional priority of improving student and school outcomes. As such, these principals indicated that utilizing collective efficacy was an integral part of the strategic planning process.

Expanding Planning Network. Nine of the 10 principals expressed that they utilized collective efficacy in strategic planning by expanding the strategic planning network to include community stakeholders. For a complete list of the participants' responses regarding expanding their planning network, please review the codebook presented in Table A-4 in Appendix A.

P2 and P4 included stakeholders in important strategic planning meetings and committees. For example, P2 said, "There is a stakeholder representative of every group on our Strategic Planning Committee" (P2), whereas P4 recounted:

The process I employ includes weekly meetings with key stakeholders.... Collective efficacy in my school community will allow for a cohesive vision to systematically improve student outcomes. Such an approach will eliminate pockets of greatness and increase student achievement exponentially by working together. (P4)

Therefore, P2 and P4 approach strategic planning and management by including community stakeholders in regular meetings, thereby utilizing collective efficacy to their advantage.

Similar to P4, both P3 and P5 included a wide range of community stakeholders in strategic planning meetings. P3 described:

Over time, I have come to expand the collective efficacy in hopes that the more hands/minds involved, the more widespread the breadth of work would become. I purposefully create multilayered committees that staff pick to focus on areas of interest to them. The hope is that they will hold each other accountable because they are genuinely interested and invested in that area. (P3)

Like the other principals, P3 included multiple types of community stakeholders on all committees in an effort not only to increase the chances of success but also to build collective efficacy. P3 also employed a unique strategy compared to the other principals interviewed in this study—namely, that the staff chooses the areas of focus in the meetings. This is an important strategy for promoting collective efficacy with respect to strategic management because the staff is invested in choosing topics they believe are important for the students. P5 also included a

wide range of individuals with different specialties in the strategic planning process. P5 said,
"Strategic planning and decision making occur in formal and informal meetings. There are
weekly leadership team meetings that include school leadership and instructional coaches.
Student and staff support meetings occur biweekly" (P5). As such, P5 also contributed unique
ideas that were not discussed by other principals, including the notion that staff and students
need support meetings. These actions would also serve to promote and increase the collective
efficacy of the school community. In essence, the principals each described ways in which they
utilized collective efficacy in strategic planning, but they also noted the use of collective efficacy
in strategic planning, in turn promoting further collective efficacy. Thus, these results indicated
that the use of collective efficacy promotes the strategic planning process, which, in turn,
promotes collective efficacy.

Subsection Two: Perceived Benefits of Collective Efficacy

Role of Collective Efficacy in Decision-Making

This section now examines the principals' perceptions of how collective efficacy alters their decision-making processes. Four themes emerged from the analysis of this question (Table 6). A summary of these themes is presented in Table 6. The first theme is that collective efficacy did not affect the decision-making process, as is discussed in Subsection Three which examines the perceived challenges associated with collective efficacy and decision making.

Another theme is that some of the principals made goals with collective efficacy in mind. The remaining two themes are: collective efficacy allows for a team orientation, and trust and transparency.

 Table 6

 Principals' Perceptions of How Collective Efficacy Affects Their Decision-Making

| How Perceptions of Collective Efficacy Affect Decision-Making | | | | | |
|---|--------------------------------------|--|--|--|--|
| Theme | Participants | Excerpt | | | |
| Collective Efficacy Does Not Drive Decisions | P10 | "It doesn't. Whether or not the school community is on the same page or not. Improvement needs to take place. The reality of it is that it falls on my shoulder as the lead to make sure the process is implemented and actually happens. While the goal is to have a shared vision, it is not always reality" (P10). | | | |
| Make Goals with Efficacy in Mind | P5, P8 | "As a leader, I value collective efficacy as I recognize that it is the glue that allows a school in need of improvement the opportunity to move forward. Trust and confidence in the leaders, the goals, and the mission of the organization are paramount" (P8). | | | |
| Team Orientation | P1, P2, P3, P4, P6, P7, P8, P9 | "As a school administrator, I always make team-informed decisions based on data. The teams' inputs are always considered before making any decision. All decisions are based on probabilities to yield higher students, stronger parental support, and the development of high-performing teachers" (P6). | | | |
| Truth and Transparency | P1, P8 | "These teams meet with fidelity. We remain transparent with our data and where we are as a school" (P1). | | | |

Principals Make Goals with Efficacy in Mind. Two of the principals (P5 and P8), in direct contrast to P10, expressed that they make goals and decisions with efficacy in mind. For example, P5 said, "It is always important for goals to be realistic and not seem too far-fetched to ensure that collective efficacy remains as high as possible" (P5). P5 demonstrated that the goals and decisions made regarding school improvement are done with collective efficacy at the forefront, so that collective efficacy can be utilized to achieve the stated goals. P8 guided their team in a similar manner, saying:

As a leader, I value collective efficacy as I recognize that it is the glue that allows a school in need of improvement the opportunity to move forward. Trust and confidence in the leaders, the goals, and the mission of the organization are paramount. (P8)

P8, like P5, expressed that effective decisions and goals require that collective efficacy be considered. Importantly, P8 recognized that collective efficacy is "glue," meaning that without it, the organization would lack structure and support. As such, P8 emphasized the importance of using collective efficacy when making decisions.

Team Orientation. The majority of the participants reported that they utilize collective efficacy when making decisions because it promotes the school community to be team-oriented and inclusive. P2 recognized that others may have alternative solutions to complex problems. P2 said, "I believe the answers are in the room, and not 'only' one person or group knows it all. It is important to do this hard work together." Similarly, P3 expressed that making decisions alone has not been effective in the past. P3 said:

I now believe heavily in the idea of collective efficacy after years of silo planning but expecting collective execution. The processes became misunderstood and disjointed, leading to limited overall success. The collective efficacy allows the community to develop, execute and refine from one lens. (P3)

P3's response is interesting in that it clearly demonstrated that in the past, P3 tried to make decisions on their own, but expected the collective to execute those plans and decisions, which ultimately was not proven effective. P3 also implied that since moving to utilize collective efficacy to make decisions, the school's situation has improved. P5 also emphasized the importance of team-oriented decision making by saying:

As a school administrator, I always make team-informed decisions based on data. The teams' inputs are always considered before making any decision. All decisions are based on probabilities to yield higher student [achievement], stronger parental support, and the development of high-performing teachers. (P6)

This response by P5 has a subtle difference from the responses of P2 and P3. P5 emphasized the importance of making *team-informed* decisions, which are not necessarily the same as team decisions. P5 appeared to be saying that, like P10, they are the sole decision maker, but unlike P10, P5 takes the information derived from the collective into consideration when making decisions for the school.

P1, P7, and P9 emphasized the importance of having multiple stakeholders present in the decision-making process. P1 reported having a team for almost every aspect of school decision making. As P1 said, "There is a team for every aspect of how we do school. [We have] an operations team, a student support team, a leadership team, vertical teacher teams, horizontal teacher teams, and a school improvement panel." In this way, P1 emphasized that the different collaborative efforts of the teams are critical for their decision-making process. Similarly, P7 said, "I always accept and garner feedback from stakeholders. I'm open to a collaborative initiative to yield the intended results and I do not work in a silo." Therefore, like P1, P7 discussed the importance of the collective perspective on important decisions, noting that it is important to get feedback from stakeholders. P9 echoed these thoughts, saying, "It is important to allow the stakeholders to be involved in the decision-making. We have open house sessions, and while I keep the school's vision at the forefront, I also value the input from the team." Thus, for many of the principals, utilizing collective efficacy in the decision-making process

strengthens the team-oriented nature of the school community, which, in turn, positively feeds back to promote collective efficacy.

Truth and Transparency. Two of the principals (P1 and P8) emphasized that making decisions utilizing collective efficacy allows for truth and transparency in the decision-making process, which they believe is important for the further development of collective efficacy. For example, P8 said:

We openly shared [the results of the teacher survey data] at staff meetings in an effort to be transparent and to continue to build our rapport with staff on how we are going to address deficient areas. (P8)

Interestingly, P8 demonstrated that the decision-making process is based on staff-reported data. This, in essence, is utilizing collective efficacy to make decisions. Such actions not only show the teachers that they are valued members of the community, but they also promote more collective efficacy. P1 expressed similar thoughts, saying, "[Our] teams meet with fidelity. We remain transparent with our data and where we are as a school." This is an important theme because recent studies have shown that transparency can promote the collective efficacy of education-based groups (Sun et al., 2019). Therefore, by being truthful and transparent, the principals not only enhance the decision-making process, but they also promote the development of further collective efficacy.

Current State of School Collective Efficacy: Perceived Benefits

This section examines the perceived benefits and positive outcomes associated with the current state of school collective efficacy. The first themes were discussed in Subsection One.

The theme of *Collective Efficacy Results in Positive Effects* is now discussed. Refer to Table 3 for all responses to the question on the *State of School Collective Efficacy*.

Collective Efficacy Results in Positive Effects. Five of the principals (P1, P2, P4, P7, and P8) indicated that the collective efficacy being built at their schools was having a positive effect on teachers, students, and support staff. P1 focused on the positive effect on teachers, describing a system where teachers stayed with the same students for multiple years. P1 said:

Teachers needed to understand the benefits of this strategy and be open to moving with those same students to the next grade level. This strategy worked and increased ELA scores in the building right before the pandemic started and we moved to remote learning. To that end, strategic collective efforts are happening to ensure that strategic plan goals are being met. (P1)

In this way, P1 demonstrated that when teachers believe in a system or a strategy, their perceived collective efficacy grows, which, in turn, positively impacts student achievement. P2 and P7 both described positive effects, but at the school level rather than the teacher level. P2 said, "It takes a village. The school goal is shared and individually, the buy-in overall is good. This helps the team work effectively" (P2). Similarly, P7 reported that collective efficacy gives "power to produce and effect change" (P7). Thus, P2 and P7 highlighted that at the school level, the collective efficacy of everyone involved has a positive impact on the school community and, most importantly, on the students. Finally, P8 saw the changes they made to their school as having a positive impact on the entire community. P8 said:

I am now in my third year at my school, and I am proud that although there were negative perceptions about our school's performance in my first year, we have worked hard to build trust again with all stakeholders.... While building trust and confidence in a school is iterative, we have made strides in the building. (P8)

P8 described the efforts of collective efficacy as reaching everyone in the school building, as well as community stakeholders, including parents and local businesses. Indeed, numerous researchers have shown that improving the collective efficacy of schools improves and strengthens neighborhoods and communities (Goddard et al., 2004; Higgins & Hunt, 2016). Taken together, the participants described their efforts towards collective efficacy as having a positive impact on their school community, including on teachers (P1), on the entire school (P2 and P7), and on the community at large (P8).

Principals Generally Make Purposeful Decisions Based on Perceived Collective Efficacy

In a similar survey question, the researcher asked the participants whether they made purposeful decisions based on collective efficacy. In general, the answer was a resounding yes. The themes that emerged from participants who said "Yes" were similar to the themes derived from the last question, indicating that data did research saturation. Specifically, the themes of team orientation and utilizing collective efficacy to maintain morale emerged from the analysis of the survey question (Table 7). Importantly, two principals (P2 and P10) noted that some or all of their decisions were not based on collective efficacy (Table 7). These negative responses to collective efficacy are discussed in Subsection Three.

Yes, Purposeful Decisions with Collective Efficacy Enhance Team Orientation. Five participants (P1, P3, P4, P8, and P9) reported that they do make purposeful decisions regarding collective efficacy. These participants discussed that the reason for these decisions was to promote the team-oriented nature of the school community. For example, P3 simply said, "Yes. I have. As mentioned previously, I intentionally structure groups that can work toward one collective end." Thus, P3 demonstrated that purposeful decisions are made to maintain and promote collective efficacy. P4 similarly commented, "Yes, given all the competing priorities, it

 Table 7

 Principals' Decision-Making Is Generally Based on Collective Efficacy

| Purposeful Decisions Made Regarding Collective Efficacy? | | | | | |
|--|-----------------------|---|--|--|--|
| Theme | Participant | Excerpt | | | |
| No, Collective Efficacy Hinders Decision-Making | P2, P10 | "No, my school's climate does not allow me to practice collective efficacy with the school community. Collective efficacy is based on collaboration and a joint understanding. That partnership does not exist in my school. While we try to foster that kind of environment, it just has not happened. So, any decisions are made in spite of the collective" (P10). | | | |
| Yes, Team Orientation | P1, P3, P4, P8, P9 | "Yes, I have made purposeful decisions in terms of the hiring of staff, partnerships, and spending of funds on programs and exposure opportunities for my students. Collective efficacy plays a huge role in this because I realized that people needed someone or something to believe in. The process, when communicated, fully resonated with so many" (P9). | | | |
| Yes, To Maintain Morale | P5 | "Yes, I have had to at times to ensure that morale and confidence among staff remain high" (P5). | | | |

Importantly, P4 added that making purposeful decisions with collective efficacy in mind will advance school-wide goals, which will, in turn, lead to school improvement as well as increase collective efficacy. P8 added to these thoughts, saying:

Yes, as we develop new initiatives, we constantly think about how we, the leadership team, will garner buy-in from all stakeholders. To illustrate, we prepared to order new texts to support small group instruction and independent reading. Three years ago, we invited our staff and students to participate in a book-tasting event where they could sample a variety of books and elect which texts they wanted to see in their classrooms. (P8)

P8 gave a concrete example of a purposeful decision that was made to promote collective efficacy. In this example, P8 involved the students and the staff in the decision-making process regarding what textbooks they wanted in the classroom. In essence, this decision brought staff and students together in a collective, collaborative event that promoted the collective efficacy of the school, thereby increasing the team-oriented nature of the school community.

Yes, Purposeful Decisions with Collective Efficacy Maintain Morale. Only one participant (P5) noted that they make purposeful decisions with collective efficacy to maintain the morale of the school community. Since this was a unique and novel answer to the question, it warrants further discussion. As shown in Table 7, P5 indicated that they made some decisions with collective efficacy to maintain the morale and confidence of the staff. Recent studies in sports have shown that increasing the confidence of the players increases the collective efficacy of the team (Chow & Fultz, 2007; Munroe-Chandler & Hall, 2004). Similar results were seen in other non-sports-related groups (Goncalo et al., 2010). Importantly, this increase in collective efficacy is linked to an increase in team performance, in sports (Leo et al., 2016; Myers et al., 2004), and in academic achievement (Hoy et al., 2002). Moreover, increasing the morale of teachers leads to increased teacher confidence (Mackenzie, 2007). Therefore, increasing morale increases confidence, and increased confidence leads to increased collective efficacy, which, in turn, increases performance.

Subsection Three: Challenges Associated with Collective Efficacy as a

Means Towards School Improvement Efforts

Current State of School Collective Efficacy: Perceived Challenges

The principals were asked about the current state of their school's collective efficacy; some principals had positive perceptions (discussed in Subsection One), while some principals

discussed challenges associated with their current state of school collective efficacy (Table 3). These specific challenges included: (a) the school had to shift its mindset, (b) improved collective efficacy led to positive effects in the schools, and (c) the school's collective efficacy was a work in progress.

Mindset Shift. Two of the principals (P1 and P9) believed that the state of collective efficacy in their schools required a mindset shift by all members of the school community. Still referencing the *needs improvement* rating of their school, P1 said, "There is a strong urge to be devoid of the 'school in need of improvement' label. Thus, we worked hard on mindset shifting" (P1). Similarly, P9 simply said, "It's a mindset shift." P1 and P9 were indicating that they were leading their school to have a growth mindset, which is important for embracing the work necessary to promote positive change. Indeed, the literature has suggested that a growth mindset on the part of teachers improves teacher self-efficacy and, in turn, student outcomes (Lin et al., 2022). Other research has suggested that when principals focus on a growth mindset, collective efficacy follows (Nordick, 2017). As such, this shift in mindset described by P1 and P9 may lead to an increase in the collective efficacy of their respective schools.

Work in Progress. Interestingly, P8 touched on this theme earlier, saying that "building trust and confidence in a school is an iterative process," highlighting that collective efficacy is a consistent work in progress. P6 and P9 echoed the opinion of P8. P6 said:

The state of collective efficacy at my school can be described as a work in progress, where most stakeholders see the need to improve and support most of the changes implemented, but some tend to give up when immediate results are not evident or a challenge/setback arises. (P6)

Importantly, P6 highlighted that setbacks prove to be difficult for community stakeholders, who become discouraged when improvements are not made immediately. Indeed, this response appears to be consistent with the culture of instant gratification that is prevalent in all levels of society in the present day (Biesta, 2019). Research has shown that communication and mentoring can effectively combat the instant gratification culture in school-based settings (Tondreau & Barnes, 2022), suggesting that the principals could increase communication with staff and community stakeholders, as well as provide mentoring services to teachers to increase school collective efficacy.

Impact of Needs Improvement Rating on Collective Efficacy

Having now defined collective efficacy according to the principals' perceptions, the researcher now turns toward the impact of the *needs improvement* rating on the school's collective efficacy. The researcher designed this question to address SQ1a and SQ1b. The majority of participants reported that the *needs improvement* rating had a negative impact on the school, students, and staff, highlighting some of the challenges of collective efficacy (SQ1b). The subthemes under the negative effect theme include: (a) assumptions that parents and staff do not care about the children, (b) the rating results in low enrollment and lack of retention, (c) there is a lack of trust in the school, and (d) the rating puts stress on the staff. By contrast, some staff members reported the needs improvement rating had a positive impact on the school, students, and staff, which highlighted some of the benefits of collective efficacy in the school setting (SQ1a). Some staff described a generally positive effect, while others described the school community as having a common goal.

 Table 8

 Principals' Perceptions of the School's Needs Improvement Rating

| Impact of Needs Improvement School Designation | | | | |
|--|--|--------------|--|--|
| Theme | Subtheme | Participants | Excerpt | |
| Negative Effect | Assumption Parents, Staff Don't Care | P1, P8, P10 | "That there would be a negative impact on the community. This would be because of the negative assumptions that this designation would bring. It is the assumption that the school is not up to par and that the school was not providing the students with adequate services and support" (P10). | |
| | Lack of Enrollment, Low Retention | P1, P7 | "The label makes parents feel as if they are making a better choice for their child by transferring them to either a higher performing school in the district or a charter school. It becomes a ripple effect of low enrollment and/or high mobility from parents taking their child out but then re-enrolling them at a later time (later that year or the following year)" (P1). | |
| | Lack of Trust | P6, P7 | "At the beginning of my career as an administrator at the school, we noticed the lack of trust and support from the internal and external school community. Most parents, business owners, teachers, and students did not believe the administration was at the school to the state. They were defensive and did not buy in until they started seeing small changes" (P6). | |
| | Stress on Staff | P3, P7, P9 | "The designation has provided minimal opportunities for highly effective teachers to become interested in our community as a career choice. We have unfortunately been restricted in spending funds" (P9). | |
| Positive Effect | | P4, P6 | "The strategic plan outlines the development of all school-related activities" (P4). | |
| | Common Goal | P2, P5 | "I believe our school goals are impactful. It gives us a common language and common footing to begin with. It defines our focus" (P2). | |

Needs Improvement Rating: Negative Effects on Schools. The majority of participants indicated that when their school was labeled with a *needs improvement* rating, the rating had a negative effect on the school, students, and staff, as well as the collective efficacy of the school. Four general themes emerged from the analysis of the data, each of which will be discussed in turn.

Assumption That Parents and School Staff Do Not Care. Three of the principals (P1, P8, and P10) indicated that the needs improvement rating led the community to the assumption that the parents and school staff did not care about their children's academic progress. For example, P1 said, "[The rating] lends itself to the assumption that the staff does not care. It lends itself to the assumption that the parents do not care and/or are not involved in their child's education." In this way, P1 argued that the rating has a negative impact on collective efficacy in that the community does not have faith in the school or the parents. Consequently, without the community's support, collective efficacy would decrease. P8 echoes these thoughts, saying:

My first year at my school, the community saw the school as a 'failing school.' Although the administrative team had a beautiful vision and goals to move our school forward, we constantly faced negative comments/views from the parent community about the poor performance and attention to students, parents, and staff concerns. (P8)

P8's response was slightly different from that of P1 in that P1 described the community as not contributing to the schools' collective efficacy, whereas P8 described the parents as not contributing. Together, these results implied that the *needs improvement* labeling of schools has a negative impact on the collective efficacy of the school community.

Lack of Enrollment and Lack of Retention. Two of the participants (P1 and P7) highlighted that the *needs improvement* rating of schools actually decreases the number of

families and students serviced by the school. As shown in Table 8, P1 described that the rating may cause parents to rethink whether they want their children to attend school. Some parents removed their children from school mid-year, leading to a lack of retention, whereas other parents simply chose not to enroll their children, leading to low enrollment. This negatively affects collective efficacy because low enrollment and lack of retention would effectively remove people who could contribute to collective efficacy. Moreover, this likely has a negative effect on other students and families, as their friends silently disappear from school. P7 had similar thoughts to P1, saying, "There is low enrollment and retention of students. Parents normally transfer to charter schools once they become proficient" (P7). P7 also described a situation in which the rating led to low enrollment and retention, which, again, decreased the school community that could engage in collective efficacy. Consequently, according to P1 and P7, the *needs improvement* rating had a negative effect on schools as well as collective efficacy.

Lack of Trust in School Administration. Two participants (P6 and P7) described the needs improvement rating as causing students and parents to not trust the school administration and teachers to educate their children properly. As shown in Table 8, P6 described a lack of both internal and external trust in the school community. P6 stated that many of the parents, community stakeholders, and even teachers did not trust that the school administration was taking proper action to improve the school and quality of education for the children. This lack of trust, in turn, led to a decrease in school efficacy, as teachers, parents, and students became defensive. P7 echoed the thoughts of P6, saying that "the labeling is a barrier for parents who want their kids in a successful school environment" (P7). P7, like P6, argued that the needs improvement rating caused parents to lose trust in the school, thereby leading to a decrease in collective efficacy.

No, Collective Efficacy Hinders Decision Making. Two principals (P2 and P10) said "No" to the question of whether they made purposeful decisions regarding collective efficacy. Interestingly, both said no for the same reason—namely, that collective efficacy can sometimes hinder the decision-making process. For example, P10 said:

No, my school's climate does not allow me to practice collective efficacy with the school community. Collective efficacy is based on collaboration and a joint understanding. That partnership does not exist in my school. While we try to foster that kind of environment, it just has not happened. So, any decisions are made in spite of the collective. (P10)

P10 vividly described that their school was not at the point of having collective efficacy. As such, P10 did not feel comfortable making decisions with the collective, likely due to the large amount of arguing that would ensue. As such, P10 found it easier to make decisions in a silo, without utilizing collective efficacy. Similarly, P2 indicated that it is sometimes difficult to be productive when using collective efficacy. P2 said, "Sometimes it is easier to get things done without so many hands in the pot." While P10 appeared to avoid utilizing collective efficacy to avoid argumentation and conflict, P2 sometimes avoided utilizing collective efficacy to increase productivity.

Collective Efficacy Does Not Drive Decision Making. Only one principal reported that collective efficacy did not drive their decision making, but their opinion was so different from the other principals that it was noteworthy of its own theme. P10 reported that collective efficacy did not drive the decision-making process, saying:

It doesn't. Whether or not the school community is on the same page or not.

Improvement needs to take place. The reality of it is that it falls on my shoulders as the

lead to make sure the process is implemented and actually happens. While the goal is to have a shared vision, it is not always [a] reality. (P10)

Participant 10 highlighted an important point—namely, that sometimes it is easier and more efficient to make decisions on one's own rather than always involving other individuals in the decision-making process. Importantly, different people may have different ideas regarding what decisions should or should not be made. This is the essence of collective efficacy. However, P10 appeared to highlight that collective efficacy can sometimes halt progress. If everyone disagrees about the correct course of action, decisions may never be made and, consequently, goals may never be attained. As such, it is important to note that collective efficacy may not always be the answer. Instead, strategic and laser-focused decision making on the part of the leader may be appropriate at times.

Negative Perceptions. Perhaps not surprisingly, the majority of principals reported that the *needs improvement* rating led the teachers and students to have negative perceptions about their schools. For example, P7 spoke about the impact on staff:

Staff doesn't like the status. It is debilitating. Some students feel less than others and feel they don't have to push themselves because the future is set. They struggle with a growth mindset. Some are resilient, and they push themselves even further. (P7)

Importantly, P7 highlighted two important ideas. First, students and staff have the opposite of a growth mindset, believing that their future success is fixed and cannot be changed. Second, this effect is likely the opposite of what the New Jersey Department of Education designed for the rating. It is doubtful that the New Jersey Department of Education meant for the rating to debilitate the staff and instill a sense of hopelessness in the students. Unfortunately, other students expressed similar feelings to their principals. For example, P6 said, "Many of my

students have eloquently shared that attending a school with a lower performance rating than all of the schools in the district is unfair because they know they are smart and talented students." Similar to P7, P6 described the *needs improvement* rating as having a negative impact on the psyche of the students.

P8, P9, and P10 vividly described the negative effects the *needs improvement* rating has on teachers. P8 recounted:

My first year, I vividly remember sitting at Organization Day and seeing our school being placed last on the list during the Convocation Ceremony. It made all of us feel terrible.

Some staff felt saddened by the school's placement, despite their efforts. (P8)

P8 highlighted another important point—namely, that the *needs improvement* rating does not accurately reflect the level of time and effort the staff and administration put into the school and the students. This impacts the staff in a negative way, diminishing their collective efficacy and morale. P10 also highlighted the negative impact on the staff, saying, "Some staff internalizes it as an indicator of their lackluster performance." As such, P10 described a similar situation to P8, demonstrating that the *needs improvement* rating genuinely diminishes the effort teachers and administrators put into their respective schools.

P9 echoed a different point that is reminiscent of an answer from P3 in Theme 3. When speaking about the impact of the *needs improvement* designation, P3 spoke about outside evaluators being present on school grounds, which, in turn, increased stress on staff (Appendix A, Table A-3). Similarly, in response to this question, P9 said, "It can be difficult at times due to the number of people who visit and are involved in the improvement process. We also must report to the state regularly. The teachers are often overwhelmed with all of the checkboxes." P9 appeared to be indicating that collective efficacy is difficult with outside evaluators on the school

campus, especially when staff and administrators feel as if they are always looking over their shoulders. Importantly, for this group of principals, the *needs improvement* rating had a drastic negative impact on the students' and teachers' perceptions of themselves, of each other, and of the school community.

Impact of Needs Improvement Rating

When asked about the impact of the needs *improvement rating* on collective efficacy, as well as the rating's impact on students, staff, and the community, the results were genuinely mixed. With respect to collective efficacy, some principals reported that the rating had a positive impact on collective efficacy, with the major theme emerging from the analysis that the school community benefited from rallying behind a common goal. This result partially addressed SQ1a, highlighting the value of collective efficacy. However, the majority of the participants reported that the *needs improvement* rating negatively affected collective efficacy, highlighting themes such as lack of enrollment, low retention of students, lack of trust in the school administration, and faculty and community assumptions that the parents and staff do not care about the school. These results partially addressed SQ1b, highlighting some of the challenges of collective efficacy.

When asked about the impact of the *needs improvement* rating on staff and students, the principals again reported mixed results. Some reported that the rating promoted collective efficacy as the staff, students, and the community rallied around the school. This further highlighted the value of collective efficacy as well as its potential to mediate change. Other principals described a similar positive effect at the beginning of the school year, which waned as the school year progressed. This suggested that collective efficacy is sometimes hard to sustain, thereby indicating a challenge associated with it. Still, other principals reported that many of the

staff, students, and community members viewed the *needs improvement* rating negatively, which inhibited the promotion of collective efficacy.

Summary

In Chapter 4, the researcher presented the results derived from the research study. The chapter began with a review of the research questions employed in the study. This section was divided into three subsections in line with the research questions. The principals reported mixed results with respect to the collective efficacy at their schools. Some reported high collective efficacy, others reported low collective efficacy, and yet others reported that collective efficacy was a work in progress. The role and impact of the *needs improvement* rating on collective efficacy in the principals' schools were also examined. Again, these questions were met with mixed opinions, but the majority of the principals reported that the *needs improvement* rating negatively impacted the collective efficacy of the school community. Additionally, the researcher examined the role of collective efficacy in the strategic planning and decision-making processes of the principals as well as the role of collective efficacy in identifying interventions. Many of the principals reported that collective efficacy played an important role in strategic management as well as in identifying interventions. Some of the principals reported using collective efficacy in the decision-making process, but others reported that the collective was sometimes unproductive for making decisions. However, the general consensus among the principals is that collective efficacy plays an integral role in the school improvement process. These results are further discussed in the context of social cognitive theory and the theory of mindset in the next chapter. Moreover, the researcher discusses implications for practice and suggestions for future research.

CHAPTER 5: RESULTS AND DISCUSSION

Summary

To date, it is presently unknown what the role of the perception of the collective efficacy of the school community is in mediating positive school improvements, especially in schools labeled as needing improvement. This study attempted to explore these perceptions of principals. During the semi-structured interviews for the present study, the researcher asked participants about their perceptions of the role of collective efficacy in a variety of tasks, including goal making, strategic management, decision making, and identifying interventions. The researcher also investigated the principals' perceptions of collective efficacy, both in general and in their schools. For the most part, the principals believed that collective efficacy positively affected the different tasks and processes utilized to modulate school improvement.

Contextual Analysis of Themes

The findings add to the research on collective efficacy and principals' perceptions of collective efficacy. Specifically, the principal consensus is that collective efficacy is integral to the school improvement process. However, some principals identified challenges with the practice and implementation of collective efficacy. These findings supported past studies indicating that collective efficacy can positively influence educational goal making (Glassman et al., 2021). Additionally, the findings showed that robust and sound leadership from principals and administrators can positively influence the perception of the collective efficacy of schools. As a result, the school has low collective efficacy, as neither pupils nor teachers support their teachers.

Similarly, Srlie and Torsheim (2012) found that a lack of organizational structure led to a deterioration in the collective efficacy of the company. Consequently, for many administrators,

leveraging collective efficacy in the decision-making process increases the team-oriented nature of the school community, which positively feeds back to increase collective efficacy. Therefore, by being truthful and transparent, principals not only improve the decision-making process but also foster the growth of greater collective efficacy. Overall, the participants viewed their efforts toward collective efficacy as having a good impact on their school community, including instructors (P1), the entire school (P2 and P7), and the community at large (P8). The findings that collective efficacy can be difficult to maintain and may necessitate a shift in the school's mindset indicated a challenge linked with it. However, some principals indicated that a significant number of staff, students, and community members viewed the needs improvement ranking unfavorably, hindering collective efficacy development.

Research Question 1

What are principals' perceptions of community collective efficacy as a means of school improvement?

Influence of Collective Efficacy on Decision Making, Strategic Planning, and Identifying Interventions

Many of the principals saw immense value in utilizing collective efficacy in decision making, strategic planning, and identifying interventions. With respect to strategic management, 90% of the principals interviewed in the study indicated that expanding their planning network facilitated positive outcomes (Chapter 4, Table 5). That is, when the principals included staff and community stakeholders, including parents, in the strategic planning and goal-making process, schools saw improvement. This indicated that collective efficacy can positively influence goal making in educational settings, in concordance with the literature findings (Glassman et al., 2021). Importantly, this is reminiscent of studies in which the empowerment of individuals led to

the empowerment of the group, thereby promoting psychological well-being (Bagci et al., 2018). Within the context of social cognitive theory, individual self-efficacy can lead to more resilient, sustainable, and determined students (Bandura, 2001). Self-efficacy, in turn, can lead to collective efficacy (Cansoy & Parlar, 2018), Moreover, the literature has indicated that collective goal making is positively correlated with increased group performance (Mulvey & Klein, 1998). Therefore, the principals' choice to utilize collective efficacy in strategic management and goal making will likely allow for school improvements.

Many, but not all, of the principals reported that utilizing collective efficacy in making important decisions for the school resulted in positive school outcomes (Chapter 4, Table 7). Some principals reported that they made decisions with collective efficacy in mind. However, the majority reported that utilizing collective efficacy in decision making promoted the team orientation of the school community. In essence, collective efficacy begets collective efficacy. These data suggested that the more principals involve the collective, the more the collective will be inspired to promote school and community growth. Indeed, increasing stakeholder buy-in and involvement is positively correlated with school reform efforts (Fuller et al., 2018; Garnett et al., 2020). Still, other literature has advised that school administrators consider the timing of soliciting stakeholder buy-in because incorrect timing can actually hinder school reform (Foote et al., 2019). The principals also reported that the use of collective efficacy in decision making increased the truth and transparency of the school's decision making. Notably, transparency in education, particularly with the distribution of educational resources, has become a somewhat contentious topic in education. Some researchers believe that some school leaders' lack of transparency regarding educational resources is akin to taking democracy out of education (Faubert, 2019). Indeed, many scholars, as well as the general public, insist on transparency and,

therefore, accountability in the K-12 education system (West et al., 2018). As such, utilizing collective efficacy in school-based decision making satisfies some of the principals whom the general public examines with respect to school reform.

Finally, a majority of principals believed that collective efficacy was critical in the process of identifying interventions. Importantly, many of the principals reported that identifying interventions and monitoring progress was a team-oriented event, involving multiple individuals to help the children needing remediation more effectively (Chapter 4, Table 4). Importantly, the principals expressed that the involvement of a small group of individuals with different skills enhanced the effectiveness of the intervention. That is, utilizing collective efficacy to identify interventions positively affects the intervention outcome. Importantly, this is congruent with the literature, which overwhelmingly indicated that utilizing collective efficacy to identify educational interventions can result in increased student well-being and mental health (García-Carrión et al., 2019). Thus, it appears that, in identifying interventions, teams should be formed to look holistically at each student's situation and outcomes.

Sub-question 1a

How do principals describe the value/possibility of community collective efficacy?

Mindset for Change, or Growth Mindset

Many positive themes recurred throughout the results section that described the value and possibility of community collective efficacy. One of the most prominent themes was that the schools took on a mindset for change, or a growth mindset, in direct concordance with Dweck's theory of the mindset. Recall that mindset theory describes the phenomenon whereby individuals who believe that certain skills and traits can be developed, as opposed to being innate, are more likely to find success in these endeavors (Dweck, 2006). This is referred to as a growth mindset.

Mindset theory argues that an individual's self-perception can profoundly affect one's accomplishments. The principals applied a growth mindset to their entire school community. In believing that the community can accomplish goals, the principals promoted collective efficacy, and the collective, in turn, sought higher goals, thereby raising the limits for what they believed could be achieved. It is important to note that collective thinking in groups is not always positive. For example, in a mob mentality, individuals sometimes take on characteristics of a large group of people that are not characteristic of each individual (Chiantera-Stutte, 2018). As such, collective efficacy, when turned negative, can hinder the growth mindset of a community. However, when viewed positively, collective efficacy can work in a positive fashion.

Growth mindset was emphasized by the principals in a number of different contexts. When asked about the state of collective efficacy at their schools, some principals reported that their schools underwent a mindset shift (Chapter 4, Table 3), and other principals suggested that collective efficacy was a work in progress (Table 3). Both of these findings are consistent with Dweck's mindset theory. In undergoing a mindset shift, the schools had to leave their *status quo* mindset and move towards a growth mindset so they could mediate the school improvement process. Indeed, a national experiment demonstrated that a growth mindset led to positive student achievement (Yeager et al., 2019). Similarly, considering a goal as a work in progress is consistent with a growth mindset, as it indicates an individual's propensity to think about growth (Murphy & Reeves, 2019).

A mindset for change, or a growth mindset, was also referenced extensively by the principals in response to the impact of the *needs improvement* rating on the students, staff, and overall school community (Chapter 4, Table 8). The *needs improvement* rating is discussed extensively in SQ1b below. However, the results from this line of questioning were generally

negative, indicating that the *needs improvement* rating has a negative effect on students, staff, and collective efficacy (Table 8). However, the staff and students taking a mindset for change after receiving the *needs improvement* rating is one positive derivation from the school rating. As stated above, a growth mindset allows for goals to be pliable and constantly moving towards a higher target. Importantly, the growth mindset of individuals can promote the growth mindset of a group. This has been shown to be the case in classrooms, where teachers transfer their growth mindsets to the students (Smith et al., 2018b). Similarly, the growth mindset of students can be contagious for teachers (Dweck, 2015). As such, the growth mindset resulting from the *needs improvement* rating can increase the growth mindset of the entire school community, resulting in an increased sense of collective efficacy.

Increased Collective Efficacy, Increased Inclusion

Many of the principals reported that when the collective efficacy at their schools is high, students and parents are actively engaged in school activities. There are many positive outcomes when students, parents, and community stakeholders participate and are actively engaged in school activities. First, this creates a greater sense of community, which could promote diversity, equity, and inclusion in the school community. Researchers studying the relationship between collective efficacy and inclusion found that collective efficacy facilitates inclusion, and vice versa (Lyons et al., 2016). Thus, it is imperative that teachers be fluent in DEI initiatives, as well as how to increase inclusion in the classroom. Strikingly, it has been reported that secondary school teachers are the least likely to implement DEI principals and initiatives in the classroom (Pozas et al., 2020). Therefore, it stands to reason that the promotion of DEI initiatives by the school administration may lead to greater inclusion in the classroom, thereby raising collective efficacy.

Sub-question 1b

How do principals describe the challenges of community collective efficacy?

The Needs Improvement School Rating

One of the biggest challenges associated with community self-efficacy in schools needing improvement is the *needs improvement* rating that the New Jersey Department of Education gives to schools. Interview Questions 2 and 4, for which results are shown in Tables 4 and 5, respectively, examined the principals' perceptions regarding the impact of the needs improvement ratio on the school community. The results from these questions were overwhelmingly negative. The principals reported that the *needs improvement* rating led to a variety of issues, including the community's assumption that the staff and parents are not invested in the education of the children (Chapter 4, Table 8); lack of enrollment and difficulties with retention (Table 8); lack of trust of the community in the school administration and staff (Table 8); increased stress on staff (Table 4); and negative perceptions of students and staff regarding themselves (Chapter 4, Table 6).

Many of the negative situations mentioned by the principals are self-perpetuating feedback cycles that only serve to decrease collective efficacy. For example, the principals reported that, due to the *needs improvement* rating, students and staff have negative perceptions of themselves. The literature has suggested that negative self-perceptions can lead to low self-esteem (Baumeister, 1993). Groups with individuals characterized by low personal self-esteem, in turn, have low group self-esteem, a property that inhibits the growth of collective efficacy (De Cremer & Oosterwegel, 1999). Therefore, transitivity might indicate that the negative self-perceptions of students and staff would lead to decrease collective efficacy. Decreased collective efficacy, then, leads to further negative perceptions of group members (Kirk & Matsuda, 2011).

Importantly, these findings are consistent with Bandura's social cognitive theory, which posits that self-efficacy is the driving force that compels people to achieve more than their perceived abilities (Donohoo & Katz, 2017). In this case, negative perceptions lead to low self-efficacy, which leads to lower levels of achievement.

As another example, some principals reported that the *needs improvement* rating caused a large amount of stress on the staff, including teachers. Teacher stress can actually transfer to students via what Bakker and Schaufeli (2000) called a contagion effect. Increased teacher stress has also been associated with poor student outcomes (Herman et al., 2018; Madigan & Kim, 2021). Indeed, when viewed through the lens of Bandura's social cognitive theory, high stress can lead to teachers with low self-efficacy (Jerusalem & Mittag, 1995), which can, in turn, affect student achievement (Bandura, 2001). Low student achievement is associated with low levels of collective school efficacy (Goddard et al., 2017; Tschannen-Moran & Barr, 2004). Therefore, high amounts of teacher stress can decrease the academic performance of students, leading to low collective efficacy. Then, low collective efficacy, in turn, leads to higher amounts of teacher stress (Klassen, 2010), perpetuating the cycle.

Collective Efficacy Can Hinder Effective Decision Making

While most of the principals indicated that collective efficacy generally promotes their decision-making process, some principals highlighted the notion that collective efficacy can actually hinder the decision-making process. At least one participant indicated that collective efficacy did not drive their decision-making process (Chapter 4, Table 6) and collective efficacy hinders decision making in general (Chapter 4, Table 7). This is an important finding because it highlights a drawback of collective efficacy, especially if all parties do not have the same goals or vision in mind. Participant 10 and, at times, Participant 2, expressed that utilizing collective

efficacy was similar to having too many cooks in the kitchen. In particular, P10 shared that the administrators and teachers at their school were not always in agreement regarding goals and visions. This led to arguing over the correct course of action, which, in turn, hindered the decision-making process of the principal. Indeed, previous studies have indicated that poor conflict management strategies of educational administrations lead to decreased efficacy of the educational organization as a whole (Balay, 2006). Importantly disagreements between teachers, administrators, and school districts are an important cause of principals choosing to leave the profession (Alenezi, 2020). Differences in opinions between administrators and teachers can also lead to teacher turnover (Redding et al., 2019). Therefore, when teachers and administrators do not share a common vision or goal, collective efficacy suffers, as does the school environment.

Limitations

One limitation of the study is that it was conducted solely within the Central Public School district. The researcher intended for participants to be drawn from multiple counties in New Jersey. However, the only participants who volunteered for the study that met the inclusion criteria were from Central Public Schools. Therefore, the actual target population of the study is Central Public School principals who lead schools in need of improvement. This may impact the transferability of the study to other counties in New Jersey; furthermore, the findings of the study may not be generalizable to other schools in other states. However, the researcher believes that the results of the study are transferable to any urban school district in the United States, as research as shown that most urban schools experience the same challenges and barriers (Howard & Milner, 2021).

Another limitation of the study is the sample size. The researcher was only able to purposefully recruit 10 participants for the study. Low numbers of participants can lead to a lack

of data saturation. However, in this case, the researcher believes that data saturation was reached. In some cases, 80-90% of the participants highlighted ideas within the same theme (see the codebook in Appendix A).

A third limitation of the study is that the survey was done via email rather than using inperson interviews. The researcher was not present when the participants were completing the
surveys. Therefore, the researcher was unable to read and interpret the participants' nonverbal
responses, including body language. As such, it is impossible for the researcher to make
suppositions based on the participants' nonverbal communication.

Finally, the paper-based survey also served as another important limitation. By confining the interview to a specific set of questions on paper, the researcher was unable to ask follow-up questions when needed. However, this limitation was, in part, mitigated by the fact that the participants took care in explaining their thoughts and did so thoroughly.

Implications for Future Practice

One important implication of this research is that Departments of Education should consider rethinking the public labeling of schools. Members and staff of schools that are in need of improvement suffer from the stigma of being associated with a poorly performing school. Indeed, the literature has a plethora of studies demonstrating that labeling schools stigmatizes not only the school but also its members (Murillo & Flores, 2002). Indeed, some researchers have argued that the stigma associated with poorly performing schools is the equivalent of shame-based motivation (Bowen & Trivett, 2014). The researcher, therefore, recommends that Departments of Education turn toward a softer way of labeling schools, when appropriate, as to not hinder the collective efficacy of the school. The rating could, perhaps, be changed from *needs improvement* (New Jersey), *F* (Florida), or *unsatisfactory* (South Carolina) to *needs*

support. This name change could serve two important purposes. First, it could focus on a positive, namely support, rather than the negative *needs improvement*. Second, a school in need of support could need a variety of types of support, including academic, financial, maintenance, or architectural. Such a label would not indicate what type of support is needed, thereby lessening the stigma associated with needing improvement. Moreover, a *needs support* rating may encourage community members to support their local school, thereby increasing the collective efficacy of the school community.

Another implication for practice is the critical importance of the growth mindset. Growth mindsets in individuals and in groups can serve to lift and motivate other individuals to achieve more than they believe possible (Dweck, 2006). Indeed, the growth mindset of a group promotes community collective efficacy, especially in education (Hanson et al., 2016). The results of the present study suggested that collective efficacy, in turn, promotes positive school change and improvement. Therefore, adding courses in professional educational education that teach the growth mindset could be helpful to all future school leaders, teachers, and support staff.

Therefore, the researcher recommends that preparation programs for educators include such courses in different mindset practices, including growth mindset, which may address challenge associated with collective efficacy. Then, future educators would have an opportunity to bring a growth mindset to their schools, thereby influencing other teachers and students. Such actions could increase the level of collective efficacy of all schools, not just schools in *need of improvement*.

Previous studies have shown that high student self-efficacy and teacher self-efficacy lead to positive changes in the school environment as well as school reform. This study was unique in that it evaluated the perceived effect of collective efficacy on the school environment and school

reform. Given that the findings indicated that increased collective efficacy can positively affect school performance and improvement, the researcher suggests that a course on how to promote self-efficacy and collective efficacy be added to professional development courses for school leaders as well as to teacher preparation programs. Perhaps educating teachers, administrators, and future school leaders regarding the positive effects of community collective efficacy would allow these leaders to bring collective efficacy to their schools and classrooms, thus becoming essential mediators of school improvement.

Suggestions for Future Research

An interesting notion is the idea of collective efficacy promoting inclusion. An avenue of future research could investigate whether increasing the DEI initiatives of K-12 schools leads to differences in student self-efficacy and teacher self-efficacy, which would, in turn, impact collective efficacy. It would be interesting to compare the findings of this study with another study that examined the state of school community efficacy throughout the states, including rural areas and throughout the country. Such data may give national educators, as well as state and local educators, a better understanding of the actual state of collective efficacy in our nation's schools. This would, in turn, have implications for positive social change regarding school improvement and school reform.

Summary

This qualitative study examined the perceptions of principals regarding the role of community collective efficacy in mediating the school improvement process. It found that the principals generally viewed collective efficacy to be positive in providing for school improvement. Generally, the study found that the principals utilized collective efficacy in all important aspects of educational leadership, including strategic management, goal making,

decision making, and identifying interventions. The collective appears to be greatly important in allowing the school to have a common goal: allowing students, staff, and the community to rally around the school. This environment, in turn, promotes the growth mindset needed for school improvement. Principals' use of collective efficacy, however, was contingent on their years of experience in an administrative role and number of years at their current schools. Principals who were employed at the same school over the course of their career were highly committed to building collaborative culture and aligning practice to the school's mission and vision. Sustaining practices such as weekly professional learning communities (PLCs), professional development, and continuous data analysis was a priority to improved student achievement and instruction. Those new to the role of the principal shared that they relied heavily on their districts for support and training on collective efficacy and school transformation. Principals new to the role spent more time working to identify and embed targeted professional development and codify systems and structures. The use of collective efficacy appears to be greatly important in allowing the school to have a common goal: allowing students, staff, and the community to rally around the school. This environment, in turn, promotes the growth mindset needed for school improvement. Regardless of who is reading this dissertation—a student, a teacher, a principal, or a community stakeholder—it is the researcher's profound hope that the reader will be open to the positive message of the study: Promote collective efficacy in all of your educational pursuits, as this will uplift those around you and, ultimately, uplift your community. We all have the power and ability to be sources of inspiration for our communities by promoting the collective above ourselves.

References

- Alenezi, A. S. (2020). Risk of turnover among US principals based on personal and school characteristics. *Journal of Research in Education*, 29(1), 1-19. https://files.eric.ed.gov/fulltext/EJ1247608.pdf
- Bagci, S. C., Turnuklu, A., & Bekmezci, E. (2018). Cross-group friendships and psychological well-being: A dual pathway through social integration and empowerment. *British Journal of Social Psychology*, *57*(4), 773-792. https://doi.org/10.1111/bjso.12267
- Bakker, A. B., & Schaufeli, W. B. (2000). Burnout contagion processes among teachers 1.

 **Journal of Applied Social Psychology, 30(11), 2289-2308. https://doi.org/10.1111/j.1559-1816.2000.tb02437.x*
- Balay, R. (2006). Conflict management strategies of administrators and teachers. *Asian Journal of Management Cases*, *3*(1), 5-24. https://doi.org/10.1177/097282010500300103
- Bandura, A. (1993). Perceived self-efficacy in cognitive development and functioning. *Educational Psychologist*, 28(2), 117-148. https://doi.org/10.1207/s15326985ep2802_3

 Bandura, A. (1997). *Self-efficacy: The exercise of control*. W. H. Freeman.
- Bandura, A. (2001). Social cognitive theory: An agentic perspective. *Annual Review of Psychology*, 52(1), 1-26. https://doi.org/10.1146/annurev.psych.52.1.1
- Bandura, A., & Walters, R. H. (1977). Social learning theory (Vol. 1). Prentice-Hall.
- Barnett, B. G., & Stevenson, H. (2016). Leading high poverty urban schools. In S. Clarke & T. O'Donoghue (Eds.), *School leadership in diverse contexts* (pp. 23-42). Routledge.
- Bass, B. M., & Avolio, B. J. (1993). Transformational leadership and organizational culture. *Public Administration Quarterly*, 112-121.

- Baumeister, R. F. (1993). Understanding the inner nature of low self-esteem: Uncertain, fragile, protective, and conflicted. In *Self-esteem: The puzzle of low self-regard* (pp. 201-218). Springer. https://doi.org/10.1007/978-1-4684-8956-9_11
- Berman, J. D., McCormack, M. C., Koehler, K. A., Connolly, F., Clemons-Erby, D., Davis, M. F., ... & Curriero, F. C. (2018). School environmental conditions and links to academic performance and absenteeism in urban, mid-Atlantic public schools.
 International Journal of Hygiene and Environmental Health, 221(5), 800-808.
 https://doi.org/10.1016/j.ijheh.2018.04.015
- Bernecker, K., & Job, V. (2019). Mindset theory. In K. Sassenberg & M. L. W. Vliek (Eds.), Social psychology in action (pp. 179-191). Springer. https://doi.org/10.1007/978-3-030-13788-5 12
- Biesta, G. (2019). What kind of society does the school need? Redefining the democratic work of education in impatient times. *Studies in Philosophy and Education*, *38*(6), 657-668. https://doi.org/10.1007/s11217-019-09675-y
- Blackwell, L., Trzesniewski, K., & Dweck, C. S. (2007). Implicit theories of intelligence predict achievement across an adolescent transition: A longitudinal study and an intervention.

 Child Development, 78, 246-263. https://doi.org/10.1111/j.1467-8624.2007.00995.x
- Bottiani, J. H., Duran, C. A., Pas, E. T., & Bradshaw, C. P. (2019). Teacher stress and burnout in urban middle schools: Associations with job demands, resources, and effective classroom practices. *Journal of School Psychology*, 77, 36-51. doi:10.1016/j.jsp.2019.10.002
- Bowen, D. H., & Trivitt, J. R. (2014). Stigmas without vouchers: The impact of ratings on student achievement in Florida's low-performing schools. Education Policy Analysis Archives.

- Boyce, J., & Bowers, A. J. (2018). Toward an evolving conceptualization of instructional leadership as leadership for learning: Meta-narrative review of 109 quantitative studies across 25 years. *Journal of Educational Administration*, 56(2), 161-182 https://doi.org/10.1108/JEA-06-2016-0064
- Burgoyne, A. P., Hambrick, D. Z., Moser, J. S., & Burt, S. A. (2018). Analysis of a mindset intervention. *Journal of Research in Personality*, 77, 21-30. https://doi.org/10.1016/j.jrp.2018.09.004
- Burić, I., & Kim, L. E. (2020). Teacher self-efficacy, instructional quality, and student motivational beliefs: An analysis using multilevel structural equation modeling. *Learning and Instruction*, 66, 101302. https://doi.org/10.1016/j.learninstruc.2019.101302
- Butina, M. (2015). A narrative approach to qualitative inquiry. *Clinical Laboratory Science*, 28(3), 190-196. http://clsjournal.ascls.org/content/28/3/190
- Butzin, S. M. (2007). NCLB: Fix it, don't nix it. *Phi Delta Kappan*, 88(10), 768. https://doi.org/10.1177/003172170708801014
- Byrd-Blake, M., Afolayan, M. O., Hunt, J. W., Fabunmi, M., Pryor, B. W., & Leander, R. (2010). Morale of teachers in high poverty schools: A post-NCLB mixed methods analysis. *Education and Urban Society*, *42*(4), 450-472. https://doi.org/10.1177/0013124510362340
- Camacho, D. A., & Parham, B. (2019). Urban teacher challenges: What they are and what we can learn from them. *Teaching and Teacher Education*, 85, 160-174. https://doi.org/10.1016/j.tate.2019.06.014
- Candela, A. G. (2019). Exploring the function of member checking. *The Qualitative Report*, 24(3), 619-628. doi:10.46743/2160-3715/2019.3726

- Cansoy, R., & Parlar, H. (2018). Examining the relationship between school principals' instructional leadership behaviors, teacher self-efficacy, and collective teacher efficacy. *International Journal of Educational Management*, *32*(4), 550-567. https://doi.org/10.1108/IJEM-04-2017-0089
- Carney, M. B., Brendefur, J. L., Thiede, K., Hughes, G., & Sutton, J. (2016). Statewide mathematics professional development: Teacher knowledge, self-efficacy, and beliefs. *Educational Policy*, 30(4), 539-572. https://doi.org/10.1177/0895904814550075
- Chen, Y., Zhou, X., & Klyver, K. (2019). Collective efficacy: Linking paternalistic leadership to organizational commitment. *Journal of Business Ethics*, *159*(2), 587-603. https://doi.org/10.1007/s10551-018-3847-9
- Chiantera-Stutte, P. (2018). Mob, people, crowds, and masses: Mass psychology and populism. *The Tocqueville Review*, 39(1), 157-176. https://doi.org/10.3138/ttr.39.1.157
- Chow, G. M., & Feltz, D. L. (2007). Exploring new directions in collective efficacy and sport.

 In M. R. Beauchamp (Ed.), *Group dynamics in exercise and sport psychology:*Contemporary Themes (pp. 239-266). Routledge.
- Creswell, J. W., & Creswell, J. D. (2018). Research design: Qualitative, quantitative, and mixed methods approaches (5th ed.). Sage.
- Darling-Hammond, L., Bae, S., Cook-Harvey, C. M., Lam, L., Mercer, C., Podolsky, A., & Stosich, E. L. (2016). *Pathways to new accountability through the Every Student Succeeds Act*. Learning Policy Institute.
- De Cremer, D., & Oosterwegel, A. (1999). Collective self-esteem, personal self-esteem, and collective efficacy in in-group and outgroup evaluations. *Current Psychology*, *18*(4), 326-339. https://doi.org/10.3102/0013189X033003003

- Donohoo, J. (2018). Collective teacher efficacy research: Productive patterns of behaviour and other positive consequences. *Journal of Educational Change*, *19*(3), 323-345. https://doi.org/10.1007/s10833-018-9319-2
- Donohoo, J., & Katz, S. (2017). When teachers believe, students achieve. *The Learning Professional*, 38(6), 20-27. www.learningforward.org
- Dunn, L., & Ambroso, E. (2019). *Balancing act: State and district roles in school improvement under ESSA*. Center on School Turnaround at WestEd.
- Dweck, C. (2015). Carol Dweck revisits the growth mindset. *Education Week*, 35(5), 20-24. https://www.edweek.org/leadership/opinion-carol-dweck-revisits-the-growth-mindset/2015/09
- Dweck, C. (2016). What having a "growth mindset" actually means. *Harvard Business Review*, 13(2), 2-5. https://hbr.org/2016/01/what-having-a-growth-mindset-actually-means
- Elmore, R. F. (2008). *Leadership as the practice of improvement*. OECDiLibrary. https://doi.org/10.1787/9789264039551-en
- Faubert, B. C. (2019). Transparent resource management: Implications for leadership and democracy in education. *International Journal of Educational Management*, *33*(5), 965-978. https://doi.org/10.1108/IJEM-02-2018-0066
- Foote, K., Henderson, C., Knaub, A., Dancy, M., & Beichner, R. (2019). Try, try again: The power of timing and perseverance in higher education reform. *Change: The Magazine of Higher Learning*, *51*(1), 50-57. https://doi.org/10.1080/00091383.2019.1547082
- Ford, T. G., Olsen, J., Khojasteh, J., Ware, J., & Urick, A. (2019). The effects of leader support for teacher psychological needs on teacher burnout, commitment, and intent to leave.

 **Journal of Educational Administration, 57(6). doi:10.1108/JEA-09-2018-0185*

- Frankel, R. M., & Devers, K. J. (2000). Study design in qualitative research—1: Developing questions and assessing resource needs. *Education for Health*, *13*(2), 251. doi:10.1080/13576280050074534
- Fuller, T., Braun, M., & Chiu, Y. L. (2018). Increasing worker buy-in for child welfare reform: Examining the influence of individual, organizational, and implementation factors.

 Children and Youth Services Review, 93, 301-306. https://doi.org/10.1016/j.childyouth.
 2018.08.003
- García-Carrión, R., Villarejo-Carballido, B., & Villardón-Gallego, L. (2019). Children and adolescents mental health: A systematic review of interaction-based interventions in schools and communities. *Frontiers in Psychology*, 10, 918. https://doi.org/10.3389/fpsyg.2019.00918
- Garnett, B., Moore, M., Kidde, J., Ballysingh, T. A., Kervick, C. T., Bedinger, L., Smith, L. C., & Sparks, H. (2020). Needs and readiness assessments for implementing school-wide restorative practices. *Improving Schools*, 23(1), 21-32. https://doi.org/10.1177/1365480219836529
- Glassman, M., Kuznetcova, I., Peri, J., & Kim, Y. (2021). Cohesion, collaboration and the struggle of creating online learning communities: Development and validation of an online collective efficacy scale. *Computers and Education Open*, 2, 100031. https://doi.org/10.1016/j.caeo.2021.100031
- Goddard, R. D., & Goddard, Y. L. (2001). A multilevel analysis of the relationship between teacher and collective efficacy in urban schools. *Teaching and Teacher Education*, *17*(7), 807-818. https://doi.org/10.1016/S0742-051X(01)00032-4

- Goddard, R. D., Hoy, W. K., & Hoy, A. W. (2004). Collective efficacy beliefs: Theoretical developments, empirical evidence, and future directions. *Educational Researcher*, *33*(3), 3-13. https://doi.org/10.3102/0013189X033003003
- Goddard, R. D., Skrla, L., & Salloum, S. J. (2017). The role of collective efficacy in closing student achievement gaps: A mixed methods study of school leadership for excellence and equity. *Journal of Education for Students Placed at Risk (JESPAR)*, 22(4), 220-236. https://doi.org/10.1080/10824669.2017.1348900
- Goncalo, J. A., Polman, E., & Maslach, C. (2010). Can confidence come too soon? Collective efficacy, conflict and group performance over time. *Organizational Behavior and Human Decision Processes*, 113(1), 13-24. doi:10.1016/j.obhdp.2010.05.001
- Greckhamer, T., & Cilesiz, S. (2022). The extension of strategic management discourse into public education administration: A critical analysis. *Organization*, 29(4), 619-652. https://doi.org/10.1177/1350508420956323
- Guest, G., Namey, E., & Chen, M. (2020). A simple method to assess and report thematic saturation in qualitative research. *PloS One*, *15*(5), e0232076. doi:10.1371/journal. pone.0232076
- Guilfoyle, C. (2006). NCLB: Is there life beyond testing? Educational Leadership, 64(3), 8.
- Hajovsky, D. B., Chesnut, S. R., & Jensen, K. M. (2020). The role of teachers' self-efficacy beliefs in the development of teacher-student relationships. *Journal of School Psychology*, 82, 141-158. Doi:10.1016/j.jsp.2020.09.001
- Haller, A., Hunt, E., Pacha, J., & Fazekas, A. (2016). Lessons for states: The "Every Student Succeeds Act" (ESSA) increases focus on and investment in supporting principal preparation and development. Center for the Study of Education Policy.

- Hanson, J., Ruff, W. G., & Bangert, A. (2016). Investigating the relationship between school level and a school growth mindset. *Journal of Educational Issues*, 2(2), 203-221. https://scholarworks.montana.edu/xmlui/handle/1/12655
- Harrison, D. D. (2021). An investigation of principals' use of stakeholder feedback for school improvement. Doctoral dissertation, Sage Graduate School. ProQuest DissertationsPublishing, 28773172
- Hennink, M., & Kaiser, B. N. (2021). Sample sizes for saturation in qualitative research: A systematic review of empirical tests. *Social Science and Medicine*, 114523. https://doi.org/10.1016/j.socscimed.2021.114523
- Herman, K. C., Hickmon-Rosa, J. E., & Reinke, W. M. (2018). Empirically derived profiles of teacher stress, burnout, self-efficacy, and coping and associated student outcomes. *Journal of Positive Behavior Interventions*, 20(2), 90-100. https://doi.org/10.1177/ 1098300717732066
- Herman, R., Gates, S. M., Arifkhanova, A., Bega, A., Chavez-Herrerias, E. R., Han, E., ... & Wrabel, S. L. (2017). *School leadership interventions under Every Student Succeeds Act:*Evidence review. The Rand Corporation. https://www.rand.org/pubs/research_reports/
 RR1550-3.html
- Higgins, B. R., & Hunt, J. (2016). Collective efficacy: Taking action to improve neighborhoods.

 NIJ Journal, 277, 18-21.
- Hodge, M., & Welch, J. (2016). An implementation perspective: Relevant lessons from No Child Left Behind (NCLB) for the implementation of Every Student Succeeds Act (ESSA).

 **Journal of Ethical Educational Leadership, 3(9), 1-17.

- Howard, T. C., & Milner, H. R. (2021). Teacher preparation for urban schools. In H. R. Milner IV & K. Lomotey (Eds.), *Handbook of urban education* (pp. 195-211). Routledge.
- Hoy, W. K., Sweetland, S. R., & Smith, P. A. (2002). Toward an organizational model of achievement in high schools: The significance of collective efficacy. *Educational Administration Quarterly*, 38(1), 77-93. doi:10.1177/0013161X02381004
- Hung, M., Smith, W. A., Voss, M. W., Franklin, J. D., Gu, Y., & Bounsanga, J. (2020).
 Exploring student achievement gaps in school districts across the United States.
 Education and Urban Society, 52(2), 175-193. https://doi.org/10.1177/00131245
 19833442
- Jacob, R., Goddard, R., Kim, M., Miller, R., & Goddard, Y. (2015). Exploring the causal impact of the McREL Balanced Leadership Program on leadership, principal efficacy, instructional climate, educator turnover, and student achievement. *Educational Evaluation and Policy Analysis*, *37*(3), 314-332. https://doi.org/10.3102/016237 3714549620
- Jasper, M., & Crossan, F. (2012). What is strategic management? *Journal of Nursing Management*, 20(7), 838-846. https://doi.org/10.1111/jonm.12001
- Jerusalem, M., & Mittag, W. (1995). Self-efficacy in stressful life transitions. In A. Bandura (Ed.), *Self-efficacy in changing societies* (pp. 177-201). Cambridge University Press.
- Johnson, L. L., Bryan, N., & Boutte, G. (2019). Show us the love: Revolutionary teaching in (un)critical times. *The Urban Review*, 51(1), 46-64. https://doi.org/10.1007/s11256-018-0488-3

- Kainz, K. (2019). Early academic gaps and Title I programming in high poverty, high minority schools. *Early Childhood Research Quarterly*, 47, 159-168. https://doi.org/10.1016/j.ecresq.2018.08.012
- Kasalak, G., & Dagyar, M. (2020). The relationship between teacher self-efficacy and teacher job satisfaction: A meta-analysis of the Teaching and Learning International Survey (TALIS). *Educational Sciences: Theory and Practice*, 20(3), 16-33. https://eric.ed.gov/?id=EJ1261816
- Kirby, M., & DiPaola, M. (2011). Academic optimism and community engagement in urban schools. *Journal of Educational Administration*, 49(5), 542-562. doi:10.1108/09578231111159539
- Kirk, D. S., & Matsuda, M. (2011). Legal cynicism, collective efficacy, and the ecology of arrest. *Criminology*, 49(2), 443-472. https://doi.org/10.1111/j.1745-9125.2011.00226.x
- Klassen, R. M. (2010). Teacher stress: The mediating role of collective efficacy beliefs. *Journal of Educational Research*, 103(5), 342-350. https://doi.org/10.1080/00220670903383069
- Knight, D. S. (2017). Are high-poverty school districts disproportionately impacted by state funding cuts? School finance equity following the Great Recession. *Journal of Education Finance*, *43*(2), 169-194. https://www.jstor.org/stable/45093658
- Lazarus, A. (2004). Reality check: Is your behavior aligned with organizational goals? *Physician Executive*, 30, 50-52.
- Lee, M., & Louis, K. S. (2019). Mapping a strong school culture and linking it to sustainable school improvement. *Teaching and Teacher Education*, 81, 84-96. https://doi.org/10.1016/j.tate.2019.02.001

- Leo, F. M., Gonzalez-Ponce, I., Sanchez-Oliva, D., Amado, D., & Garcia-Calvo, T. (2016).

 Exploring direction between cohesion and collective efficacy and relationships with performance of football teams. *South African Journal for Research in Sport, Physical Education and Recreation*, 38(3), 113-126.
- Lin, W., Yin, H., & Liu, Z. (2022). The roles of transformational leadership and growth mindset in teacher professional development: The mediation of teacher self-efficacy.

 Sustainability, 14(11), 6489. doi:10.3390/su14116489
- Linares, L., Herrero-Fernández, D., Iriarte, L., & Estévez, A. (2017). Psychometric properties from a Spanish validation of depressive experiences questionnaire (DEQ). *Acción**Psicológica, 14(1), 137-149. http://urn.kb.se/resolve?urn=urn:nbn:se:hj:diva-43678
- Lincoln, Y. S., & Guba, E. G. (1990). Judging the quality of case study reports. *International Journal of Qualitative Studies in Education*, 3(1), 53-59. https://doi.org/10.1080/0951839900030105
- Liu, S., & Hallinger, P. (2018). Principal instructional leadership, teacher self-efficacy, and teacher professional learning in China: Testing a mediated-effects model. *Educational Administration Quarterly*, *54*(4), 501-528. https://doi.org/10.1177/0013161X18769048
- Liu, Y., Bellibaş, M. Ş., & Gümüş, S. (2021). The effect of instructional leadership and distributed leadership on teacher self-efficacy and job satisfaction: Mediating roles of supportive school culture and teacher collaboration. *Educational Management Administration and Leadership*, 49(3), 430-453. https://doi.org/10.1177/174114
- Locke, E. A., & Schattke, K. (2019). Intrinsic and extrinsic motivation: Time for expansion and clarification. *Motivation Science*, *5*(4), 277. doi:10.1037/mot0000116

- Lyons, W. E., Thompson, S. A., & Timmons, V. (2016). 'We are inclusive. We are a team. Let's just do it': Commitment, collective efficacy, and agency in four inclusive schools.

 *International Journal of Inclusive Education, 20(8), 889-907. https://doi.org/10.1080/13603116.2015.1122841
- Lysova, E. I., Allan, B. A., Dik, B. J., Duffy, R. D., & Steger, M. F. (2019). Fostering meaningful work in organizations: A multi-level review and integration. *Journal of Vocational Behavior*, 110, 374-389. https://doi.org/10.1016/j.jvb.2018.07.004
- Mackenzie, N. (2007). Teacher morale: More complex than we think? *The Australian Educational Researcher*, 34(1), 89-104. doi:10.1007/BF03216852
- Madigan, D. J., & Kim, L. E. (2021). Does teacher burnout affect students? A systematic review of its association with academic achievement and student-reported outcomes.

 *International Journal of Educational Research, 105, 101714. https://doi.org/10.1016/j.ijer.2020.101714
- Maier, A., Daniel, J., Oakes, J., & Lam, L. (2017). Community schools as an effective school improvement strategy: A review of the evidence. Learning Policy Institute.
- Mathis, W. J., & Trujillo, T. M. (2016). Lessons from NCLB for the Every Student Succeeds Act.

 National Education Policy Center.
- Maulana, R., Opdenakker, M. C., & Bosker, R. (2016). Teachers' instructional behaviors as important predictors of academic motivation: Changes and links across the school year.

 *Learning and Individual Differences, 50, 147-156. doi:10.1016/j.lindif.2016.07.019
- Mount, A. L. (2018). *Teacher experiences in the school improvement process*. Ed.D. Doctoral dissertation, University of West Georgia. ProQuest Dissertations Publishing, 10976014.

- Mulford, B. (2003). School leaders: Challenging roles and impact on teacher and school effectiveness. Paper prepared for the OECD Improving School Leadership Activity. www.oecd.org/edu/schoolleadership
- Mulvey, P. W., & Klein, H. J. (1998). The impact of perceived loafing and collective efficacy on group goal processes and group performance. *Organizational Behavior and Human*Decision Processes, 74(1), 62-87. https://doi.org/10.1006/obhd.1998.2753
- Munroe-Chandler, K. J., & Hall, C. R. (2004). Enhancing the collective efficacy of a soccer team through motivational general-mastery imagery. *Imagination, Cognition and Personality*, 24(1), 51-67. doi:10.2190/UM7Q-1V15-CJNM-LMP4
- Murillo, E. G., & Flores, S. Y. (2002). Reform by shame: Managing the stigma of labels in high stakes testing. *Journal of Educational Foundations*, 16(2), 93.
- Murphy, M. C., & Reeves, S. L. (2019). Personal and organizational mindsets at work. *Research in Organizational Behavior*, *39*, 100121. https://doi.org/10.1016/j.riob.2020.100121
- Myers, N. D., Payment, C. A., & Feltz, D. L. (2004). Reciprocal relationships between collective efficacy and team performance in women's ice hockey. *Group Dynamics: Theory,**Research, and Practice, 8(3), 182. https://doi.org/10.1037/1089-2699.8.3.182
- Ni, Y., Yan, R., & Pounder, D. (2018). Collective leadership: Principals' decision influence and the supportive or inhibiting decision influence of other stakeholders. *Educational Administration Quarterly*, 54(2), 216-248. doi:10.1177/0013161X17723427
- Ninković, S. R., & Knežević Florić, O. Č. (2018). Transformational school leadership and teacher self-efficacy as predictors of perceived collective teacher efficacy. *Educational Management Administration and Leadership*, 46(1), 49-64. https://doi.org/10.1177/1741143216665842

- No Child Left Behind Act. (2001). No Child Left Behind Act. https://www2.ed.gov/nclb/landing.jhtml
- Nordick, S. (2017). Fundamental features of fostering teacher collective efficacy: Principals' attitudes, behaviors, and practices. Doctoral dissertation, Utah State University. All Graduate Theses and Dissertations. 6639. https://digitalcommons.usu.edu/etd/6639
- Olivier, E., Archambault, I., De Clercq, M., & Galand, B. (2019). Student self-efficacy, classroom engagement, and academic achievement: Comparing three theoretical frameworks. *Journal of Youth and Adolescence*, 48(2), 326-340. doi:10.1007/s10964-018-0952-0
- Opdenakker, M. C., Maulana, R., & den Brok, P. (2012). Teacher-student interpersonal relationships and academic motivation within one school year: Developmental changes and linkage. *School Effectiveness and School Improvement*, 23(1), 95-119. doi:10.1080/09243453.2011.619198
- Perera, H. N., Calkins, C., & Part, R. (2019). Teacher self-efficacy profiles: Determinants, outcomes, and generalizability across teaching level. *Contemporary Educational Psychology*, 58, 186-203. https://doi.org/10.1016/j.cedpsych.2019.02.006
- Poulou, M. S., Reddy, L. A., & Dudek, C. M. (2019). Relation of teacher self-efficacy and classroom practices: A preliminary investigation. *School Psychology International*, 40(1), 25-48. https://doi.org/10.1177/0143034318798045
- Pozas, M., Letzel, V., & Schneider, C. (2020). Teachers and differentiated instruction: Exploring differentiation practices to address student diversity. *Journal of Research in Special Educational Needs*, 20(3), 217-230. https://doi.org/10.1111/1471-3802.12481

- Prenger, R., & Schildkamp, K. (2018). Data-based decision making for teacher and student learning: A psychological perspective on the role of the teacher. *Educational Psychology*, 38(6), 734-752. https://doi.org/10.1080/01443410.2018.1426834
- Reaves, S. J., & Cozzens, J. A. (2018). Teacher perceptions of climate, motivation, and self-efficacy: Is there really a connection? *Journal of Education and Training Studies*, 6(12), 48-67. :10.11114/jets.v6i12.3566
- Redding, C., Booker, L. N., Smith, T. M., & Desimone, L. M. (2019). School administrators' direct and indirect influences on middle school math teachers' turnover. *Journal of Educational Administration*, *57*(6), 708-730. https://doi.org/10.1108/JEA-10-2018-0190
- Riegelsberger, J., Sasse, M. A., & McCarthy, J. D. (2005). The mechanics of trust: A framework for research and design. *International Journal of Human-Computer Studies*, 62(3), 381-422. https://doi.org/10.1016/j.ijhcs.2005.01.001
- Riley, D. L., Meredith, J., & Butler, A. N. (2019). State responsibilities and opportunities for school improvement under the Every Student Succeeds Act. Council of Chief State School Officers.
- Ross, J., & Bruce, C. (2007). Professional development effects on teacher efficacy: Results of randomized field trial. *Journal of Educational Research*, *101*(1), 50-60. https://doi.org/10.3200/JOER.101.1.50-60
- Schipper, T., Goei, S. L., de Vries, S., & van Veen, K. (2018). Developing teachers' self-efficacy and adaptive teaching behaviour through lesson study. *International Journal of Educational Research*, 88, 109-120. https://doi.org/10.1016/j.ijer.2018.01.011

- Shelton, M. L. (2020). The story of growth: A case study of a New Jersey underperforming underprivileged high school. Doctoral dissertation, Seton Hall University. ProQuest Dissertations Publishing, 27666074.
- Shin, M., & Bolkan, S. (2021). Intellectually stimulating students' intrinsic motivation: The mediating influence of student engagement, self-efficacy, and student academic support.

 *Communication Education, 70(2), 146-164. doi:10.1080/03634523.2020.1828959
- Shukla, K. D., Kuril, S., & Chand, V. S. (2020). Does negative teacher behavior influence student self-efficacy and mastery goal orientation? *Learning and Motivation*, 71, 101653. https://doi.org/10.1016/j.lmot.2020.101653
- Simpson, R. L., Lacava, P. G., & Sampson Graner, P. (2004). The No Child Left Behind Act:

 Challenges and implications for educators. *Intervention in School and Clinic*, 40(2),

 67-75. https://doi.org/10.1177/10534512040400020101
- Skaalvik, E. M., & Skaalvik, S. (2007). Dimensions of teacher self-efficacy and relations with strain factors, perceived collective teacher efficacy, and teacher burnout. *Journal of Educational Psychology*, 99(3), 611. doi:10.1037/0022-0663.99.3.611
- Smith, J. G., Morin, K. H., Wallace, L. E., & Lake, E. T. (2018a). Association of the nurse work environment, collective efficacy, and missed care. *Western Journal of Nursing Research*, 40(6), 779-798. doi:10.1177/0193945917734159
- Smith, T., Brumskill, R., Johnson, A., & Zimmer, T. (2018b). The impact of teacher language on students' mindsets and statistics performance. *Social Psychology of Education*, 21(4), 775-786. https://doi.org/10.1007/s11218-018-9444-z

- Sökmen, Y. (2021). The role of self-efficacy in the relationship between the learning environment and student engagement. *Educational Studies*, 47(1), 19-37. https://doi.org/10.1080/03055698.2019.1665986
- Sørlie, M. A., & Torsheim, T. (2011). Multilevel analysis of the relationship between teacher collective efficacy and problem behaviour in school. *School Effectiveness and School Improvement*, 22(2), 175-191. doi:10.1080/09243453.2011.563074
- Sun, A., & Xia, J. (2018). Teacher-perceived distributed leadership, teacher self-efficacy and job satisfaction: A multilevel SEM approach using the 2013 TALIS data. *International Journal of Educational Research*, 92, 86-97. https://doi.org/10.1016/j.ijer.2018.09.006
- Sun, N., Rosson, M. B., & Carroll, J. M. (2019). Promoting online learning community with identity transparency. In P. Diaz & A. Ioannou (Eds.), *Learning in a digital world: An* introduction (pp. 135-153). Springer.
- Supovitz, J. A., D'Auria, J., & Spillane, J. P. (2019). Meaningful and sustainable school improvement with distributed leadership. CPRE Research Reports. https://repository.upenn.edu/cpre_researchreports/112
- Taylor, S. J., Bogdan, R., & DeVault, M. (2015). *Introduction to qualitative research methods:*A guidebook and resource. John Wiley & Sons.
- Tondreau, A., & Barnes, Z. T. (2022). Mentoring and being mentored in an age of social media connection and instant gratification. In M. B. Mackinem, L. J. Ritter, & A. Bagasra (Eds.), *The changing faces of higher education: From Boomers to Millennials*. Vernon Press.

- Tschannen-Moran, M., & Barr, M. (2004). Fostering student learning: The relationship of collective teacher efficacy and student achievement. *Leadership and Policy in Schools*, 3(3), 189-209. https://doi.org/10.1080/15700760490503706
- Tschannen-Moran, M., & McMaster, P. (2009). Sources of self-efficacy: Four professional development formats and their relationship to self-efficacy and implementation of a new teaching strategy. *The Elementary School Journal*, 110(2), 228-245. doi:10.1086/605771
- Tsivitanidou, O. E., Gray, P., Rybska, E., Louca, L., & Constantinou, C. P. (Eds.). (2018).

 *Professional development for inquiry-based science teaching and learning. Springer International.
- Uchida, A., Michael, R. B., & Mori, K. (2018). An induced successful performance enhances student self-efficacy and boosts academic achievement. AERA Open. https://doi.org/10.1177/2332858418806198
- van Aalst, D. A., Huitsing, G., Mainhard, T., Cillessen, A. H., & Veenstra, R. (2021). Testing how teachers' self-efficacy and student-teacher relationships moderate the association between bullying, victimization, and student self-esteem. *European Journal of Developmental Psychology, 18*(6), 928-947. https://doi.org/10.1080/17405629.2021. 1912728
- Vasileiou, K., Barnett, J., Thorpe, S., & Young, T. (2018). Characterising and justifying sample size sufficiency in interview-based studies: systematic analysis of qualitative health research over a 15-year period. *BMC Medical Research Methodology*, 18(1), 1-18. https://doi.org/10.1186/s12874-018-0594-7

- Versland, T. M., & Erickson, J. L. (2017). Leading by example: A case study of the influence of principal self-efficacy on collective efficacy. *Cogent Education*, 4(1), 1286765. doi:10.1080/2331186X.2017.1286765
- Voelkel, Jr., R. H., & Chrispeels, J. H. (2017). Understanding the link between professional learning communities and teacher collective efficacy. *School Effectiveness and School Improvement*, 28(4), 505-526. https://doi.org/10.1080/09243453.2017.1299015
- Watson, G. (2006). Technology professional development: Long-term effects on teacher self-efficacy. *Journal of Technology and Teacher Education*, 14(1), 151-166.
- West, M. R., Henderson, M. B., Peterson, P. E., & Barrows, S. (2018). The 2017 EdNext poll on school reform: Public thinking on school choice, common core, higher ed, and more. *Education Next*, 18(1), 32-53.
- Wilcox, K. C., & Lawson, H. A. (2018). Teachers' agency, efficacy, engagement, and emotional resilience during policy innovation implementation. *Journal of Educational Change*, 19(2), 181-204. https://doi.org/10.1007/s10833-017-9313-0
- Wronowski, M. L. (2018). Filling the void: A grounded theory approach to addressing teacher recruitment and retention in urban schools. *Education and Urban Society*, *50*(6), 548-574. https://doi.org/10.1177/0013124517713608
- Yeager, D. S., & Dweck, C. S. (2012). Mindsets that promote resilience: When students believe that personal characteristics can be developed. *Educational Psychologist*, 47(4), 302-314. https://doi.org/10.1080/00461520.2012.722805
- Yeager, D. S., Hanselman, P., Walton, G. M., Murray, J. S., Crosnoe, R., Muller, C., ... & Dweck, C. S. (2019). A national experiment reveals where a growth mindset improves achievement. *Nature*, *573*(7774), 364-369. https://doi.org/10.1038/s41586-019-1466-y

- Zheng, X., Yin, H., & Li, Z. (2019). Exploring the relationships among instructional leadership, professional learning communities and teacher self-efficacy in China. *Educational Management Administration and Leadership*, 47(6), 843-859. https://doi.org/10.1177/1741143218764176
- Zimmerman, B. J. (1995). Self-efficacy and educational development. *Self-efficacy in Changing Societies*, 1(1), 202-231. https://doi.org/10.1017/CBO9780511527692.009
- Zimmerman, B. J., Bandura, A., & Martinez-Pons, M. (1992). Self-motivation for academic attainment: The role of self-efficacy beliefs and personal goal setting. *American Educational Research Journal*, 29(3), 663-676. https://doi.org/10.3102/0002831202 9003663

Appendix A: Codebook

Table A-1Principals' Definitions of Collective Efficacy Codebook

Definition of Collective Efficacy

| Participant | Theme | Excerpt |
|-------------|---|---|
| P1 | Works Towards a Positive Goal | "Collective efficacy is the work that happens towards a shared & collective belief that all students can learn and excel" (P1). |
| P2 | Collaboration, Teamwork Work Towards a Positive Goal | "It means that everyone works together to meet a common goal" (P2). |
| P3 | Ability | "A group's ability to produce the desired result" (P3). |
| P4 | Collaboration, Teamwork | "Collective efficacy compliments organizational systems, which are the underpinnings of collaborationCollective efficacy ensures that stakeholders work with fidelity to achieve school-wide goals to leverage the greatest impact to supersede pockets of the greatest throughout the school community" (P4). |
| P5 | Ability Collaboration, Teamwork Works Towards a Positive Goal | "Having a sense of togetherness as an organization. Using that togetherness to foster common beliefs in the ability to accomplish goals and to work effectively" (P5). |
| P6 | Ability Collaboration, Teamwork | "I would describe collective efficacy as the ability to collaborate with parents, students, and staff (instructional and non-instructional) to hire, develop and sustain staff members who are firmly committed to transforming any under performing academic institution. This process requires ongoing communication, parental support, and ongoing monitoring" (P6). |
| P7 | Works Towards a Positive Goal | "Collective efficacy is an overall product of a positive effect" (P7). |
| P8 | Ability | "I would define collective efficacy as the ability to have collective trust. [It is] the confidence individuals within an organization have towards achieving success" (P8). |
| P9 | Collaboration, Teamwork | "Collective efficacy is the belief that everyone can and will succeed" (P9). |
| P10 | Ability | "A group's shared understanding gives that group the ability to control other behaviors. It's a way to create order using a system of understandings" (P10). |

Table A-2Principals' Perceptions of the School's Needs Improvement Rating Codebook

Impact of Needs Improvement School Designation

| Participant | Theme | Subtheme | Excerpt |
|-------------|--|--|---|
| | • | Assumption Parents, Staff Don't Care | "It lends itself to the assumption that the staff does not care. It lends itself to the assumption that the parents do not care and/or are not involved in their child's education" (P1). |
| P1 | Negative Effect | Lack of Enrollment, Low Retention | "The label makes parents feel as if they are making a better choice for their child by transferring them to either a higher performing school in the District or a charter school. It becomes a ripple effect of low enrollment and/or high mobility from parents taking their child out but then re-enrolling them at a later time (later that year or the following year)" (P1) |
| P2 | Positive Effect | Common Goal | "I believe our school goals are impactful. It gives us a common language and common footing to begin with. It defines our focus" (P2). |
| P3 | Negative Effect | Stress on Staff | "Designations for improvement signal that "outside" evaluators would be coming in to oversee work and observe on a consistent basis" (P3). |
| P4 | Positive Effect | | "The strategic plan outlines the development of all school-related activities" (P4). |
| P5 | Positive Effect | Common Goal | "Having a sense of collective efficacy can help with increasing optimism. It can be a confidence builder as school personnel works to improve student achievement" (P5). |
| P6 | Negative Effect Positive Effect | Lack of Trust | "At the beginning of my career as an administrator at the school, we noticed the lack of trust and support from the internal and external school community. Most parents, business owners, teachers, and students did not believe the administration was at the school to the state. They were defensive and did not buy in until they started seeing small changes" (P6). |
| | | Positive Effect | "Caring for the school facility also played a huge role in the community. Walking around daily to greet parents, neighbors, business owners and "the young crew" that always hangs out in the school surroundings helped us gain their trust" (P6). |
| | Negative Effect | Lack of Trust | "The labeling is a barrier for parents who want their kids in a successful school environment" (P7). |
| P7 | | Lack of Enrollment, Low Retention | "There is low enrollment and retention of students. Parents normally transfer to charter schools once they become proficient" (P7). |

| | | _ | |
|-----|--------------------|--|---|
| | | Stress on Staff | "Teachers leave to higher-performing schools" (P7). |
| P8 | Negative Effect | Assumption Parents, Staff Don't Care | "My first year at my school, the community saw the school as a "failing school." Although the administrative team had a beautiful vision and goals to move our school forward, we constantly faced negative comments/views from the parent community about the poor performance and attention to students, parents, and staff concerns" (P8). |
| P9 | Negative Effect | Stress on Staff | "The designation has provided minimal opportunities for highly effective teachers to become interested in our community as a career choice. We have unfortunately been restricted in spending funds" (P9). |
| P10 | Negative Effect | Assumption Parents, Staff Don't Care | "That there would be a negative impact on the community. This would be because of the negative assumptions that this designation would bring. It is the assumption that the school is not up to par and that the school was not providing the students with adequate services and support" (P10). |

Table A-3Principals' Perceptions of the State of School Collective Efficacy Codebook

State of School Collective Efficacy

| Participant | Theme | Excerpt |
|-------------|--|--|
| | Mindset Shift | "There is a strong urge to be devoid of the "school in need of improvement" label. Thus, we worked hard on mindset shifting" (P1). |
| P1 | Resulted In Positive Effect | "Teachers needed to understand the benefits of this strategy & be open to moving w/ those same students to the next grade level. This strategy worked & increased ELA scores in the building right before the Pandemic started and we moved to remote learning. To that end, strategic collective efforts are happening to ensure that strategic plan goals are being met" (P1). |
| P2 | Resulted In Positive Effect | "It takes a village. The school goal is shared and individually, the buy-in overall is good. This helps the team work effectively" (P2). |
| P3 | Initial Positive, Then Negative | "Initially, the collective efficacy is high due to the feeling of "this is our school, and we must protect it." Over long/extended periods of time the momentum tends to wane" (P3). |
| P4 | Resulted In Positive Effect | "My school community consists of two buildings that were recently merged for the first time. This upcoming academic year requires laser-sharp focus in having the teachers and staff embark on a major shift in seeing the school community as one" (P4). |
| P5 | High Collective Efficacy | "My school community consists of two buildings that were recently merged for the first time. This upcoming academic year requires laser-sharp focus in having the teachers and staff embark on a major shift in seeing the school community as one" (P5). |
| P6 | Work in Progress | "The state of collective efficacy at my school can be described as a work in progress where most stakeholders see the need to improve and support most of the changes implemented, but some tend to give up when immediate results are not evident or a challenge/set back arises" (P6). |
| P7 | Resulted In Positive Effect | "Power to produce and effect" (P7). |
| P8 | Resulted In Positive Effect | "I am now in my third year at my school, and I am proud that although there were negative perceptions around our school's performance in my first year, we have worked hard to build trust again with all stakeholdersWhile building trust and confidence in a school is iterative, we have made strides in building" (P8). |
| P9 | Work in Progress | "While it is evident it is in pockets amongst some grade teams and individual teachers. It is a work in progress" (P9) |
| . • | Mindset Shift | "It is a mindset shift" (P9). |
| P10 | Low Collective Efficacy | "I think it is lacking in my school. That is why we are currently dealing with chaos. There is a lack of structure and a lack of confidence in the administration. It feels like everyone is out for themselves" (P10). |

Table A-4Principals' Perceptions of the Effect of the School's Rating on Staff and Students Codebook

| | Effect of Sch | nools' Needs Improvement Ratings on Students and Staff |
|-------------|---|---|
| Participant | Theme | Excerpt |
| P1 | Mindset for Change | "We work hard on climate & culture aspects. We use staff & student surveys throughout the year to keep a pulse on building morale. We have managed to make staff & students feel empoweredeven with having low state test scores. There is a fostered belief that we will move the needle & get out of status. We try to keep the work of Carol Dweck at the forefront & ensure that we are moving from a growth mindset frame of reference" (P1). |
| P2 | Positive Perception, Effect | "Our school is outperforming the local public school. Our scores are a major focus for our teachers. They come second to our primary focus, the well-being of our students" (P2). |
| P3 | Rating Has No Effect on Perceptions | "Only veteran staff actually mention a time when students performed better. New members tend to not mention ratings at all but are drawn to the location of the school in relation to where they live. Ratings overall are not a factor." (P3) |
| P4 | Mindset for Change | "In the blended conversations I have had with teachers, support staff, and the previous administration, I understand that there is a desire to improve" (P4) |
| P5 | Positive Perception, Effect | "Our school is considered a top-performing school among schools in the school district. We ranked second for math achievement and forth for ELA achievement" (P5). |
| ' | Negative Perceptions | "Many of my students have eloquently shared that attending a school with a lower performance rating than all of the schools in the district is unfair because they know they are smart and talented students" (P6). |
| Do. | Rating Has No Effect on Perceptions | "[The students] have also shared that it is simply a label and that such a label does not define who they truly are" (P6). |
| P6 | Mindset for Change | "[The students] expressed that they are committed to changing the narrative by showing on state assessments that they can perform as high as any other student in the district. Bilingual and ESL teachers take the lower performance rating more seriously. They are more competitive, believe in their students and invest more time planning/tailoring instruction and pushing students to acquire the language to properly exit out of the bilingual program" (P6). |
| P7 | Negative Perceptions | "Staff doesn't like the status. It is debilitating. Some students feel less than others and feel they don't have to push themselves because the future is set. They struggle with a growth mindset. Some are resilient, and they push themselves even further" (P7). |
| P8 | Negative Perceptions | "My first year, I vividly remember sitting at Organization Day and seeing our school being placed last on the list during the Convocation Ceremony. It made all of us feel terrible. Some staff felt saddened by the school's placement, despite their efforts" (P8). |

| | Mindset for Change | "[Staff] felt empowered to do more. [Teachers and students] felt that they could do more to move the school forward. As an administrator, my message the first few months of the year was, "this is where we are; this is not who we are," and so slowly, we began to change these sentiments" (P8). |
|-----|---|--|
| P9 | Negative Perceptions | "It can be difficult at times due to the number of people who visit and are involved in the improvement process. We also must report to the state regularly. The teachers are often overwhelmed with all of the check boxes" (P9). |
| | Negative Perceptions | "Some staff internalizes it as an indicator of their lack luster performance" (P10). |
| P10 | Positive Perception, Effect | "While some staff looks at the benefits of the additional funding as a way to get needed supplies" (P10). |
| | Rating Has No Effect on Perceptions | "The kids really don't know the difference" (P10). |

Table A-5Principals' Perceptions of Role of Collective Efficacy in Strategy Planning Codebook

| | Rol | e of Collective Efficacy in Strategic Management |
|-------------|---|---|
| Participant | Theme | Excerpt |
| P1 | Data Analysis, SMART Goals | "Being a school driven by DDI (Data Driven Instruction), we always begin with Data Analysis to determine where we are & where we need to go. Teachers are part of this process. The Leadership Team develops SMART Goals within the Strategic Plan based on the Data. As we meet with teachers to support the creation of their SGO (Student Growth Objectives), they are clear about how their respective data fits into the scope of the whole" (P1). |
| | Strategic Planning Leads to Collective Efficacy | "This process lends itself to collective efficacy" (P1). |
| P2 | Expand Planning Network | "There is a stakeholder representative of every group on our Strategic Planning Committee" (P2). |
| P3 | Expand Planning Network | "Over time, I have come to expand the collective efficacy in hopes that the more hands/minds involved, the more widespread the breadth of work would become. I purposefully create multi-layered committees that staff pick to focus on areas of interest to them. The hope is that they will hold each other accountable because they are genuinely interested and invested in that area" (P3). |
| P4 | Expand Planning Network | "The process I employ includes weekly meetings with key stakeholdersCollective efficacy in my school community will allow for a cohesive vision to systematically improve student outcomes. Such an approach will eliminate pockets of greatness and increase student achievement exponentially by working together. " (P4). |
| | Data Analysis, SMART Goals | "Additionally, teachers are given two periods a week to work collaboratively to discuss student data, analysis, and action plans to address deficiencies or maintain gains" (P4). |
| P5 | Expand Planning Network | "Strategic planning and decision-making occur in formal and informal meetings. There are weekly leadership team meetings that include school leadership and instructional coaches. Student and Staff Support meetings occur biweekly" (P5). |
| P6 | Expand Planning Network | "We form committees, and take time to norm/calibrate as a team before conducting formal observation. After goals are set we share them with the community." (P6). |
| | Data Analysis, SMART Goals | "We also pull and analyze student achievement data (including staff and student attendance) as well as teachers' observation data and resources available before setting goals for the strategic plan" (P6). |

| P7 | Expand Planning Network | " A collaborative process. Review progress and prior goals with teachers and parents. We look at it over the summer and garner feedback in PLC's. we use opportunities for professional development and further develop goals. The plan is presented to parents and community partners" (P7). |
|-----|-------------------------------|---|
| P8 | Data Analysis, SMART Goals | "As we prepared to develop strategic plans at our school, we used previous data to set SMART goals and engaged in a brainstorming discussion around how we were going to move our school forward" (P8). |
| | Expand Planning Network | "We utilized staff, parent, and student survey data to frame our conversationsWe set strategic timelines over how we would monitor each goal and allowed opportunities to revise the goal. During GLM's and PLC's, we shared our strategic plan with staff to gain input and feedback from staff" (P8). |
| | Data Analysis, SMART Goals | "We prepared to develop strategic plans in Aug of every year. We brainstorm as a team and use previous data to set goals" (P9). |
| P9 | Expand Planning Network | "In terms of collective efficacy, we involve all stakeholders by inviting them to plan with us, looking at staff surveys, and collaborating with parents" (P9). |
| P10 | Expand Planning Network | "The process begins by trying to get the stakeholders together to work together to try to get specific goals of the school. We try to get input from each stakeholder is important to ensure everyone's views have been heard and seen. After the initial engagement it is important to create a realistic plan of action, which will enable the group to be focused on the created end goals" (P10). |

 Table A-6

 Principals' Perceptions of How Collective Efficacy Affect Their Decision-Making Codebook

Collective Efficacy Affects Decision-Making

| Participant | Theme | Excerpt |
|-------------|-------------------------------------|---|
| P1 | Team Orientation | "There is a Team for every aspect of how we do schoolan Operations Team, a Student Support Team, a Leadership Team, Vertical Teacher Teams, Horizontal Teacher Teams, SIP Team (School Improvement Panel), etc" (P1). |
| | Truth and Transparency | "These teams meet with fidelity. We remain transparent with our data and where we are as a school" (P1). |
| P2 | Team Orientation | "I believe the answers in the room and not "only" one person or group knows it all. It is important to do this hard work together" (P2). |
| P3 | Team Orientation | "I now believe heavily in the idea of collective efficacy after years of silo planning but expecting collective execution. The processes became misunderstood and disjointed, leading to limited overall success. The collective efficacy allows the community to develop, execute, and refine from one lens" (P3). |
| P4 | Team Orientation | "Research confirms that gains in achievement have to be done in collaboration instead of silos. This reform strategy builds capacity through shared perspectives, pedagogy, and supportive relationships. To ensure that such measures permeate my school community, I will implement collective efficacy" (P4). |
| P5 | Make Goals with Efficacy in Mind | "It is always important for goals to be realistic and not seem too far- fetched to ensure that collective efficacy remains as high as possible" (P5). |
| P6 | Team Orientation | "As a school administrator, I always make team-informed decisions based on data. The teams' inputs are always considered before making any decision. All decisions are based on probabilities to yield higher students, stronger parental support, and the development of high-performing teachers" (P6). |
| P7 | Team Orientation | "I accept and garner feedback from stakeholders. I'm open to a collaborative initiative to yield the intended results and I do not work in a silo" (P7). |
| | Team Orientation | "During our discussions and yearly reviews, we continuously used teacher survey data to evaluate staff trust, buy-in, and confidence" (P8). |
| P8 | Truth and Transparency | "We openly shared [the results of the teacher survey data] at staff meetings in an effort to be transparent and continue to build our rapport with staff on how we were going to address deficient areas" (P8). |
| | Make Goals with Efficacy in Mind | "As a leader, I value collective efficacy as I recognize that it is the glue that allows a school in need of improvement the opportunity to move forward. Trust and confidence in the leaders, the goals, and the mission of the organization are paramount" (P8). |

| P9 | Team Orientation | "It is important to allow the stakeholders to be involved in the decision-making. We have open house sessions, and while I keep the school's vision at the forefront I also value the input from the team" (P9). |
|-----|---|---|
| P10 | Collective Efficacy Doesn't Drive Decisions | "It doesn't. Whether or not the school community is on the same page or not. Improvement needs to take place. The reality of it is that it falls on my shoulder as the lead to make sure the process is implemented and actually happens. While the goal is to have a shared vision, it is not always reality" (P10). |

Table A-7Principals' Decision-Making Are Generally Based on Collective Efficacy Codebook

Purposeful Decisions Made Regarding Collective Efficacy?

| Participant | Theme | Excerpt |
|-------------|---|---|
| P1 | Yes, Team Orientation | "Yes. Again, I, as Principal, can not do this work alone. Thus, "Think Tanks" & "Thought Partners" have to be a part of the strategy and decision-making if you are truly going to move your school forward and close academic achievement gaps" (P1). |
| P2 | No, Collective Efficacy Hinders Decision-Making | "Sometimes it is easier to get things done without so many hands in the pot" (P2). |
| P3 | Yes, Team Orientation | "Yes, I have. As mentioned previously, I intentionally structure groups that can work toward one collective end" (P3). |
| P4 | Yes, Team Orientation | "Yes, given all the competing priorities, it is critical to lead purposefully with the end goal in mind–advancing school-wide goals" (P4). |
| P5 | Yes, To Maintain Morale | "Yes I have had to at times to ensure that morale and confidence among staff remain high" (P5). |
| P6 | N/A | |
| P7 | N/A | |
| P8 | Yes, Team Orientation | "Yes, as we develop new initiatives, we constantly think about how we, the leadership team, will garner buy-in from all stakeholders. To illustrate, as we prepared to order new texts to support small group instruction and independent reading three years ago, we invited our staff and students to participate in a book-tasting event where they could sample a variety of books and select which texts they wanted to see in their classrooms" (P8). |
| P9 | Yes, Team Orientation | "Yes, I have made purposeful decisions in terms of the hiring of staff, partnerships, and spending of funds on programs and exposure opportunities for my students. Collective efficacy plays a huge role in this because I realized that people needed someone or something to believe in. The process, when communicated, fully resonated with so many" (P9). |
| P10 | No, Collective Efficacy Hinders Decision-Making | "No, my school's climate does not allow me to practice collective efficacy with the school community. Collective efficacy is based on collaboration and a joint understanding. That partnership does not exist in my school. While we try to foster that kind of environment, it just has not happened. So, any decisions are made in spite of the collective effective" (P10). |

Table A-8Principals' Perceptions of Process for Identifying Interventions Codebook

Process for Identifying Interventions

| Participant | Theme | Excerpt |
|-------------|--------------------------------------|--|
| P1 | Data-Driven Process | "We determine which teachers are the strongest at which contents. We analyze student data. Teachers create Corrective Instruction Plans based on the data. This determines which students need support with which Priority and/or Prerequisite Standards" (P1). |
| | Response to Intervention (RTI) | "We then determine what support will look like for those students. It could be various forms of RTI (Response to Intervention, Tutoring, Additional Guided Readings Sessions, Saturday Academy, etc." (P1). |
| P2 | Response to Intervention (RTI) | "We use our R.T.I system and our I&RS team to make decisions collectively" (P2). |
| P3 | Team-Driven Intervention | "Interventions are identified by a team that has direct knowledge/ interaction with said population, ability to identify root causes, analysis of work samples or data, and can offer solutions for correction. The team is also responsible for benchmark review of the effectiveness of agreed upon intervention" (P3). |
| P4 | Team-Driven Intervention | "Identifying interventions for the targeted population is done in collaboration with central office leaders, content area specialists, research-based practices, and reflection" (P4). |
| P5 | Team-Driven Intervention | "Interventions have to be planned and executed for targeted populations to improve student achievement to ensure equity. Collective efficacy is fostered to ensure that efforts are made to improve student outcomes" (P5). |
| | Data-Driven Process | "The first step is pulling and analyzing student data as a team to determine which educational areas are in need of support for each child" (P6). |
| P6 | Team-Driven Intervention | "Very frequently, the children in need of educational or socio-emotional support are referred to, then discussed/presented at I&RS meetings. Administration and instructional coaches, as well as academic interventionists, are part of this meeting and always contribute by sharing which supports can best assist the student throughout the school year. It is always communicated to the family and the teachers in hopes of supporting the student to grow academically. Professional relationships, coaching, and ongoing informal conversations with the students and staff also play a huge role throughout the process" (P6). |
| P7 | Data-Driven Process | "We look at data to determine and isolate standards that students have not mastered, and we then group them and remediate those who have not" (P7). |

| P8 | Team-Driven Intervention | "We use an Academic and Behavior Tracker school-wide to identify academic, behavioral, attendance, and disciplinary concerns. There is a sense of collective efficacy as it pertains to the decision-making behind interventions. In addition, we meet as an SST Team that involves our social workers, counselors, our Parent Liaison, Attendance Counselor, and the school administration. During these meetings, there is a sense of collective efficacy to support our students that are in need of interventions" (P8). | |
|----|-----------------------------|--|--|
| P9 | Team-Driven Intervention | "I identify interventions and programs with the support of the district, my school constituents, and research. I have a huge population of students who struggle academically. Because I believe in working together, I decided that this process was important to have all parties engaged and involved" (P9). | |
| | Data-Driven Process | "We look at behavior, academic, and assessment data to determine the proper interventions for students" (P9). | |

Appendix B: IRB Approval



August 23, 2022

Crystal Joye Seton Hall University

Re: 2022-358

Dear Crystal,

At its Summer meeting, the Research Ethics Committee of the Seton Hall University Institutional Review Board reviewed and approved your research proposal entitled, "Exploring Perception About Collective Efficacy in Urban K-8 New Jersey Schools Designated to the School Improvement Process" submitted. This memo serves as official notice of the aforementioned study's approval. Enclosed for your records are the stamped original Consent Form and recruitment flyer. You can make copies of these forms for your use.

The Institutional Review Board approval of your research is valid for a one-year period from the date of this letter. During this time, any changes to the research protocol, informed consent form or study team must be reviewed and approved by the IRB prior to their implementation.

You will receive a communication from the Institutional Review Board at least 1 month prior to your expiration date requesting that you submit an Annual Progress Report to keep the study active, or a Final Review of Human Subjects Research form to close the study. In all future correspondence with the Institutional Review Board, please reference the ID# listed above.

Thank you for your cooperation.

Sincerely,

Mara C. Podvey, PhD, OT

Associate Professor

Phyllis Hansell, EdD, RN, DNAP, FAAN

Professor

Co-Chair, Institutional Review Board
Co-Chair, Institutional Review Board

Appendix C: Informed Consent Form



Informed Consent Letter

Title of Research Study: Exploring Principals' Perceptions about Collective Efficacy in Urban K-8 New Jersey Schools Designated for the School Improvement Process

Principal Investigator: Crystal Joye

Department Affiliation: Executive Educational Leadership Management and Policy (Ed.D) Program, Seton Hall University.

Sponsor: This research is supported by the Executive Educational Leadership Management and Policy (Ed.D) Program.

Brief summary about this research study:

The following summary of this research study is to help participants decide whether or not they want to participate in the study. Participants have the right to ask questions at any time. The purpose of this qualitative narrative study is to Explore Principals Perceptions about Collective Efficacy in Urban K-8 New Jersey Schools Designated for the School Improvement Process. Within the context of the current study, collective efficacy in a school is the perception of the principal and teachers that the faculty as a whole can execute the course of action required to positively impact student achievement. Semi-structured interviews will be conducted with 10-12 elementary school principals from one school district in New Jersey. This research focuses on principal in elementary school from Kindergarten to eighth grade. The principals work in a large urban school district and will be selected because of their experience working in a comprehensive or targeted school designated for the school improvement process. The participants have varying years of experience in the principal role.

The participants will be asked to answer eight interview questions. Participants will participate in one interview and one follow-up interview.

We expect that participants will be in this research study for one hour for the initial interview and 30 minutes for the follow-up interview.

The primary risk of participation is possible loss of privacy. Only the necessary personal information will be collected and coded immediately after the interview. The data will securely be stored on a password-protected device so that only the PI can access the information. There are no known risks associated with this study. To mitigate risk, participants are encouraged to skip any questions they feel are not comfortable. Additional time/breaks will be provided to each participant if requested. And finally, each participant can withdraw themselves from the study at any point.

The main benefit of participation is to lend significant new insights pertaining to educational practice and used by theory used by principals. The perspectives collected during this research may inform new understandings of the use of efficacious practices during the school

improvement process. Knowledge gained through this research may contribute to practices and strategies that may prove valuable for schools engaging in the improvement process. Additionally, the results from the study will add to best practice in United States public schools, particularly in New Jersey.

Purpose of the research study:

Participants are being asked to take part in this research study because they have previously led or currently lead a K-8 school urban school in New Jersey identified as needing improvement. Women nor minorities will be excluded from the subject pool. Additionally, prisoners and illiterate persons will be excluded.

Participants who agree to participate in this research study will be expected to engage for one hour for the initial interview and 30 minutes for the follow up.

Participants will be one of 12 people who are expected to participate in this research study.

What you will be asked to do:

Participation in this research study will include: A pre-interview questionnaire that you will complete prior to the consent form and the virtual interview. One interview and one follow up interview. If permission is given to be audio recorded, the researcher will audio record the interviews using an iPhoneX. The interviews will take no longer than one hour virtually with the researcher only. At the start of the interview the researcher will explain the reason for the research. The interview protocol will start as follows:

- The participant will be assigned a pseudonym, which will be used during the transcription of the audio recording.
- The participant will be asked to describe a brief history of their background in education.
- The researcher will share the research questions related to the study:

RQ1. What are principals' perceptions of community collective efficacy as a means towards school improvement?

- a. How do principals describe the value/possibility of community collective efficacy?
- b. How do principals describe the challenges of community collective efficacy?

Before the interview is over, the researcher will remind you that if you wish to opt out of the study, you still have time before the researcher finalizes the study.

Contact information:

If participants have questions, concerns, or complaints about this research project, they can contact the principal investigator (Crystal Joye) at Crystal.Joye@student.shu.edu or Dr. David Reid, Department: Education Leadership, Management & Policy Email: david.reid@shu.edu Office Phone: 9732752324 or the Seton Hall University Institutional Review Board ("IRB") at 9737619334 or at irb@shu.edu.

Optional Elements:

Audio will be performed as part of the research study. Please indicate your permission to participate in these activities by placing your initials next to each activity.

| participate | e in these activities by placing your in | itials next to each activity. | |
|-------------------------|--|---|--|
| I agrec | I disagree | cord my [audio or video] interview. In | |
| understanding this is o | | done to help with data collection and analysis. not share these recordings with anyone outside of | |
| I hereby | consent to participate in this research | study. | |
| Signature | e of participant | Date | |
| Printed n | name of participant | | |
| Signature | e of person obtaining consent | Date | |
| Printed n | name of person obtaining consent | | |